# Estimating Nonseparable Preference Speci...cations for Asset Nonset Participants.

Kris Jacobs<sup>n</sup>

(0 ctaber 1999) Very Preliminary Please do not 0 uote

#### A bstract

This paper uses panel data to estimate preference speci...cations that are nonseparable in consumption and leisure. Because the econometric analysis uses panel data, it die ers from existing econometric studies that use a representative agent framework. Because the paper focuses on the nonlinear implications of the theory, it is die erent from most existing panel data studies that investigate linearizations. The evidence shows that we only dotain intuitively plausible estimates when using samples that contain households who own riskless and risky assets. For those samples, estimated parameter values are radically die erent from existing studies. The ...notings are therefore of interest to an extensive literature in macroeconomics and ...nance

JEL Classi...cation 6 12, E2, D91

Keywards:

#### 1 Introduction

The estimation of preference parameters is of interest to an extensive literature in economics and ... nance. The size and sign of these parameters is often used to address important policy questions and to judge the empirical performance of a certain model or a methodological approach. For instance, the esse with which economic agents tracle or consumption or leisure over time determines the effectiveness of a given government design of social security programs. A loop following the work by Kyolland and P rescott (1982), an extensive literature has debated the ability of "real business cycle models" to explain the data. For

 $<sup>^{12}\</sup>text{Faculty}$  of M anagement, M di iII U niversity, 1001 Sherbrooke Street W est, M ontreal, Canada H 3A 16 5; Tel: (514) 398-4025; Fax: (514) 398-3876; E-mail: Jacobse M anagementM di iII.CA .

this debate, the level of the intertemporal rate of substitution is a critical issue. A vailable estimates of the rate of intertemporal substitution, and therefore assessments of the usefulness of these models, are determined by the choice of preference speci...cation and the econometric methodology used in estimation.

W hereas the debate about real business cycle models has somewhat subsided, theuse of dynamic models has become a cornerstone of modern mear occorranics. If any of these dynamic studies evaluate theoretical models by comparing actual data with data simulated from the model under study. These simulations re quire estimates of a wide variety of parameters, including preference parameters. For this literature, reliable estimates of parameters that describe intertemporal behavior are therefore very valuable. Such estimates are also of interest for more traditional econometric tests of dynamic models. Traditional econometric testing uses uses some model implications for testing conditional on the estimated parameters. If the parameter estimates are unreliable, the resulting tests are not likely to be informative. In asset pricing a popular alternative approach to testing is often used (e.g. see M ehra and P rescott (1985) and H ansen and Jaconnathan (1991)). The implications of the theory are evaluated for diagrant values of the behavioral parameters. If parameter values can be found that can explain the theoretical implications, the researcher asks himself the question if those parameter values are intuitively plausible. Whereas the judgment of plausibility is presumably based on a wide range of available sources, including introspection, existing parameter estimates are useful as reference points.

This paper contributes to this literature by presenting estimates and test statistics dotained using nonlinear Euler equations. This estimation and test strategy is quite popular in the representative agent literature. If ansen and Singleton (1982) pioneered the approach using a time separable constant relative risk aversion (T.S-CR.R.A.) utility function. The approach was later used to analyze other preference speci...cations, and M ankiw, R otemberg and Summers (1985) (henceforth III RS) and Eicherbaum, H ansen and Singleton (1988) (henceforth EHS) provide an analysis of nonseparabilities between consumption and leisure in a representative agent context. This paper dixers from those studies because it provides an analysis of nonseparable preferences using panel data. The available panel data literature on preference estimation is extensive. If owever, many panel data studies analyze separable preferences. Those that analyze nonseparable preferences almost always investigate linearizations of the nonlinear Euler equations, and therefore parameter estimates in those papers may be difficult to relate to the ones in this paper. Because there are so many studies that use linearizations, I provide an analysis of the linearized Euler equations below using the same datasets. If ... not that for the same dataset, the di¤erence between parameter estimates from the linearized and the nonlinear equations is always substantial.

For each speci...cation, estimation and test results are reported for three dimerent samples. The ...rst sample contains all households who ful...II certain selection or teria. The second and third samples contains households who participate in asset markets, with the selection or teria for these two samples slightly dimerent. I ...nd that whereas parameter estimates are always intuitively plausi-

ble for the second and third samples, this is almost never the case for the ...rst sample. For those samples that yield intuititively plausible estimates, the re sults indicate that leisure enters the utility function in a statistically signi...cant way casting doubt on studies that use separable preference speci...cations. The parameter values are signi...cantly di¤ erent from those obtained by other panel data studies, and from the ones presented in M RS (1985) and EHS (1988). A Iso, point estimates are di¤erent from the estimates typically used in simulation studies in dynamic macroeconomics. The special case of logarithmic preferences, which is also often used in this literature, is rejected by the data The rejection of separability is also of interest for the asset pricing literature, which almost without exception uses separable preferences. Statistical tests of overidentifying restrictions indicate less evidence against the model compared to similar tests obtained using a TS-CRRA speci...cation in Jacobs (1999). It is therefore tempting to conducte that the asset pricing literature should adopt nonseparable preferences. However, it will be arqued that additional analysis is needed to determine whether nonseparable preferences can solve a number of well-known asset priding puzzles.

The paper proceeds as follows. Section 2 provides a detailed motivation for this study and an extensive discussion of related research. Section 3 discusses the data and Section 4 discusses the estimation and testing methodology. Section 5 presents the empirical results. Section 6 provides a robustness analysis of the results obtained in Section 5. Section 7 relates the ... notings of the paper to the related literature and Section 8 provides a short condusion.

#### 2 Motivation

Estimation and testing are carried out with a minimum of auxiliary assumptions. We assume the existence of a large number of individuals with an identical perperiod utility function, speci...ed as

$$u(G_{;t};I_{it}) = \frac{1}{\tilde{A}}G_{i;t}^{\tilde{A}}I_{i;t}. \tag{1}$$

where  $\mathbf{q}_{;t}$  is the consumption of individual i in period t and  $\mathbf{l}_{i;t}$  is the leisure of individual i in period t. We present estimates of the parameters using the ...rst-order condition with respect to consumption (assuming an interior solution)

$$U_{c}(G_{i:t}; I_{i:t})_{i=1} : i:t = \emptyset :$$
 (2)

where  $_{\rm ait}$  is the Lagrange multiplier associated with individual its time thought constraint 0 neway togenerate testable restrictions from the model is to specify the assets that the individual can invest in. If any existing panel data studies assume that the individual only has a riskless asset (a bond) to invest in. If ere

we assume that the individual can invest in risky assets (stocks) as well as in bands. This yields the following two Euler equations

$$1 = {^{-}E}_{t_{i}} {_{1}} \left(\frac{1}{Q_{t_{i}}}\right) \left(\frac{G_{;t}}{G_{;t_{i}}}\right)^{\tilde{A}_{i}} \left(\frac{I_{i;t}}{I_{i;t_{i}}}\right)^{\cdot} : \tag{3}$$

$$1 = {^{-}E_{t_{i} \ 1}} \left(\frac{p_{t} + \ C_{t}}{p_{t_{i} \ 1}}\right) \left(\frac{G_{;t}}{G_{;t_{i} \ 1}}\right)^{\tilde{A}_{i} \ 1} \left(\frac{I_{i;t}}{I_{i;t_{i} \ 1}}\right)^{\cdot} : \tag{4}$$

where  $p_t$  is the price of the risky asset in period t;  $q_t$  is the dividend on the risky asset in period t;  $q_t$  is the (normalized) price of the riskless asset in period t;  $t_t$  is the discount factor; and E  $t_t$  is the mathematical expectation conditional on information available at time t.

T wo important advantages of focusing on the estimation of (3) and (4) have to be emphasized at this point. First, it is dear that the derivation of (3) and (4) assumes that the household is at an interior sulution w.r.t. the holdings of the riskless and the risky asset and w.r.t. consumption. If ovever, no assumption has been made about the existence of corner solutions or rigidities in the labor market. This is important for an empirical study, because many individuals are dearly at a corner solution for leisure choice, and it is di¢ cult to tell from available data whether an individual is at a corner solution or not It would of course be preferable to include information about leisure choice in the econometric analysis, because the extra information would generate more precise parameter estimates and more powerful test statistics. If owever, given that the extra power comes at a higher risk of dassi...cation error and inconsistent estimation, the focus in this paper is on (3) and (4) in isolation. Second, to derive (3) and (4) no assumptions about the structure of ... nancial markets have to be made beyond the existence of a riskless and a risky asset. In particular, we do not have to make the complete markets assumption which underlies all of the representative agent studies of consumption and leisure choice (e.g. see Ell S (1988) and MRS (1985). In view of the overwhelming evidence appiret complete markets in the literature (see Cochrane (1991), III ace (1991) and III ayashi, A. Itonji and Kotlikor (1996), the absence of this assumption is reassuring Jacobs (1998b, 1999) shows that in the absence of a complete markets assumption, parameter estimates and test results are very diagrent from the representative agent literature for a time separable constant relative risk aversion (T.S.CR.R.A.) speci...cation.

The importance of estimating preference parameters using elementary implications of dynamic equilibrium models can hardly be overstated. Whereas economists continue to disagree about many issues, the last two decades have witnessed the emergence of dynamic aspects of human behavior as one of the cornerstones of economics. Estimates of parameters characterizing intertemporal behavior are therefore of interest to almost every theory-based economic study because often such parameters are of fundamental importance to assess the theory's policy implications or its empirical performance

B esides pointing out the general relevance of this exercise, it is also instructive to provide a more detailed motivation of its importance by discussing its impact on dia erent research areas. Such a more detailed discussion is necessary because the ostensibly simple issue of estimating behavioral parameters such as the ones in (3) and (4) has been tackled using dia erent auxiliary assumptions in dia erent areas of the economics and ...nance literature, each motivated by dia erent questions and concerns. Perhaps unsurprisingly, these dia erent approaches have given rise to dia erent parameters estimates and widely varying assessments of their relevance and policy implications.

I discuss the importance of preference speci...cation in three diameters (but related) areas ...irst, the literature on dynamic macrosconomics; second, the literature that unites macrosconomics and asset pricing and third, the literature on consumption, savings and labor economics. I his somewhat arbitrary division into research areas is motivated by the fact that within each group, the econometric and methodogical tools used in the analysis are fairly similar. First, in macrosconomics an extensive literature has developed that investigates the implications of general equilibrium models by way of simulation. In principle, this literature does not depend on estimates of behavioral parameters, but the empirical implementation used in many studies requires estimates of behavioral parameters as model inputs. Therefore, speci...c values for behavioral parameters are mostly assumed as part of the primitives of the model. For instance, Kydland and P rescott (1982) use the Cobb D oudes preference speci...cation

$$u(G_{i;t};I_{i;t}) = \frac{1}{{}^{\circ}\mu} (\mathring{G}_{i;t}I_{i;t}^{1_i})^{\mu}$$
 (5)

Kydland and P rescott (1982) argue that it is straightforward to parame terize (4) by interpreting "as the percentage of the agent's time allocated to consumption activities. A loop by interpreting  $\mathring{q}_{i,t}l_{i,t}^{1_i}$  as a generalized version of a unit of consumption, they interpret  $1_i$   $\mu$  as the rate of constant relative risk aversion. They therefore set "equal to 1/3, based on the fact that households' allocation of time to normarket activities is about twice as large as the allocation to market activities, and they use three diagreent values of the parameter  $\mu$  in their simulations: -1, -0.5 and -0.1.

It is dear that (7) is simply a rede...nition of (1) with  $\mu=\tilde{\mathbb{A}}+\cdot$  and  $^\circ=\tilde{\mathbb{A}}=\mu$ . For our purposes, it is convenient to report on the parameters in equation (1), because inspection of the standard error on - allows us to determine whether lesureenters the utility function in a signi... can the manner and whether the underlying optimization problem is concave. If one ver, for reasons of comparison with the real business cycle literature, all tables also list the point estimates of  $^\circ$  and  $\mu$  that are implied by our estimates of  $\tilde{\mathbb{A}}$  and  $\cdot$ . In this context, the parameter 1  $_{\tilde{1}}$   $\mu$  will be referred to as the rate of relative risk aversion in accordance with the terminology proposed by Kydland and P rescott (1982).

Several studies in dynamic macroeconomics employ the utility speci...cation (7) used in Kydland and Prescott (1982) (e.g. see Backus, Kehoe and Kydland

(1992)). A central question is therefore whether the range of estimates for  $\mu$  proposed by Kydland and P rescott (1982) contains the parameter estimate from the data. If oreover, many existing studies use a logarithmic speci...cation that is a special case of (7) for  $\mu=0$  (see Christiano and Eicherbaum (1992)). It is therefore of interest to verify if this restriction hads in the data <sup>3</sup>

The second area of interest that has studied the importance of leisure is the asset pricing literature. Following the work of Lucas (1978) and B reeden. (1979), the asset pricing literature has focused on estimating and testing general equilibrium models of asset pricing. Parameter estimation is critically important in this context because the theory is usually evaluated conditional on parameter estimates dotained under the maintained hypothesis that the the ary holds. A fiter several studies demonstrated the inadequacy of the T.S.-CR.R.A. speci...cation to explain the volatility of asset returns (see II ansen and Singleton (1982,1983,1984), III ehra and Prescott (1985)), several authors proposed the indusion of leisure in the utility function to remedy the problem (see III RS (1985) and EHS (1988)). However, the representative agent literature has conducted that the presence of leisure in the utility function is not likely to solve notarious asset priding puzzles. Using the intuition captured by the work of Il ansen and Jagannathan (1991), the problem with consumption based asset pricing is that the consumption series is not volatile enough to generate sufficient volatility in asset returns, given the available parameterization of the utility function. For the TS-CRRA speci...cation, one needs an "implausibly large" risk aversion pa rameter to generate sufficient variability in the intertemporal marginal rate of substitution. B ecause the time series of monthly, quarterly or yearly appreciate leisure is also very smooth, it is not surprising that the parameters needed to generate sufticient variability in the IM RS are judged implausible as well (see also Campbell, Loand III adKinlay (1997, p.326).

M are in general, the relationship between nonseparable preferences and asset prices has not attracted abundant interest in the …nance literature  $^4$  M ayers

<sup>&</sup>lt;sup>1</sup>The merits and criticisms of this literature are not the focus of this paper. Whereas the term "real business cycles" has been hotly debated, and the criginal methodology in Kiydland and Pirescott (1982) has also come under scrutiny (e.g. refer to the debate between Kiydland and Pirescott (i) and Hilansen and Hiedkman (ii), the analysis of dynamic models has become the state of the art in macroeconomics. Whereas much of this research in economic dynamics has little in common with the ideological battleground of real business cycles, many of these studies (but not all) use a simulation setup where parameter values from other studies are used as inputs into the analysis.

<sup>&</sup>lt;sup>2</sup> Following the analysis in Kydland and P rescott (1982), many of the studies that use the Cobb-D ouglas sped...cation (7) specify time nonseparabilities in leisure. Strictly speaking therefore, our estimates cannot be compared with the calibrated values used in these studies.

<sup>&</sup>lt;sup>3</sup>The literature on dynamic general equilibrium models has recognized the importance of heterogeneity. The firamework used in Rogerson (1988) and Hansen (1985) allows for the preferences of the representative agent to die enfrom those of the individual agents populating the economy. He eventheless, estimates of the preferences of individual agents remain valuable.

 $<sup>^4\</sup>text{B}$  esides the poor performance of leisure in an intertemporal priding context, there are other explanations for this. First, the emergence of arbitrage priding techniques (R oss (1976)) does not necessitate an elaborate description of the economic environment. Second, economic priding factors such as consumption have not performed well in cross-sectional asset priding (B reeden, G ibbons and L itzenberger (1989)).

(1972, 1973) emphasizes the importance of leisure and human capital accumulation at an early stage in the context of the Capital & sset P ricing M cold. Cochrane and H ansen (1992) and Fama (1991, p. 160) make a forceful point for establishing a theoretical and empirical connection between asset returns and intuitively plausible pricing factors emanating from general equilibrium models, such as leisure. In the theoretical literature, the importance of leisure choice has been analyzed among others by B cole, M enton and Samuelson (1992) and B asak (1999). The work of Jagannathan and W ang (1996) suggest that human capital accumulation could be of interest in cross-sectional asset pricing

This paper argues that the perceived failure of nonseparable utility specimations to solve asset pricing puzzles may be due to the complete markets assumption that underlies these studies. If this assumption does not hold, it is the properties of the individual agent's consumption and leisure instead of aggregate consumption and leisure that determine asset prices. If ankiw and I eldes (1991) and Jacobs (1998 b, 1999) show that these aggregation issues are critically important when evaluating the TS-CR RAL specimation. This paper investigates if similar arguments apply for the evaluation of preferences that are nonseparable in consumption and leisure

Whereas the existence of nonseparabilities between consumption and leisure has not received a lot of attention recently, the speci...cation of preferences has received ever more attention in the asset pricing literature. If our notably, the speci...cation of preferences with time nonseparabilities has become quite popular (e.g., see II, bet (1990), Campbell and Cochrane (1995), Constantinides (1990), Dietemple and I apatero (1991) and Sundaresan (1989)). If hereas this paper does not dany that this type of speci...cation is intuitively plausible, it explores howmuch leverage can be obtained from a speci...cation without time nonseparabilities once the complete markets assumption is relaxed.

This study is also related to a third important area of research. Empirical analyses of consumption choice, savings, and leisure choice are treated as one group mainly because this literature contains a large number of empirical studies that use panel data <sup>6</sup> For purposes of comparison with this study a

<sup>&</sup>lt;sup>5</sup>The stochastic properties of leisure are also of interest in the literature on nontraded assets, even though leisure does not enter the utility function in many of the studies in that literature (e.g. see D etemple (1999) and S vensson and W erner (1993)).

For panel data studies in the consumption literature, see A ttanasio and B rowning (1995), A ttanasio and W eber (1995), Keane and R unkle (1992), R unkle (1991), I eldes (1989) and the references in the excellent overview paper by B rowning and L usardi (1994). For panel data studies in the labor supply literature see A I tonji (1984), H am (1984), H edwarn and II aCurdy (1980) and III aCurdy (1981). Whereas many studies in this literature use panel data, there is an equally extensive literature that relies on aggregate data (see H all (1988) and the references in D eaton (1992) in the consumption and savings literature and A I tonji (1982) and I I tonji and II am (1990) in the labor supply literature). It is hard to determine why this particular empirical literature has started relying relatively more on panel data compared to the other research areas discussed in this section. Several authors maintain that aggregation problems make it tenuous to use time series of consumption and leisure to draw conclusions about individual preferences (e.g. see B rowning and L usardi (1996). P erhaps another reason for the focus on panel data in this literature is the extensive literature on static demand systems which predates most dynamic studies. T his literature relies almost exclusively on panel data (see D eaton and III uellbauer (1980), D eaton (1992)).

potential problem is that most existing studies analyze linearized versions of Euler equations (3) and (4) as well as linearized versions of other intertemporal Euler equations such as those implied by optimal leisure choice. The main motivation for using linearizations is probably that measurement error is an important problem with panel data (see A Itanji (1984) and A Itanji and Siow (1987)), and it is easier to deal with measurement error problems in a linear context. While this study does not dany the importance of measurement error; it takes the opposite view that it is worth investigating the original nonlinear Euler equations (3) and (4) directly. If measurement error is a serious problem in this context and for this purpose, it will likely lead to implausible parameter estimates and a lack of robustness. It turns out that for most speci... cations this is not the case.

There are other reasons why it is not necessarily straightforward to compare parameter estimates obtained from the analysis of the nonlinear Euler equations (3) and (4) with the parameter estimates in the panel data studies mentioned above. First, the link with the labor supply literature is that we focus on the point estimate and the statistical signi...cance of the parameter . . If owever, whereas we obtain estimates of this parameter using the ...rst order condition with respect for consumption, much of the labor supply literature uses the ...rst order condition with respect to leisure (e.g. seell a Curdy (1981), if Itanji (1986), if am (1986). Illustrating this methodology for our preference speci...cation (1), consider the Euler equation obtained by combining the intertemporal optimality condition and the ...rst order condition writ. I elisure

$$1 = {}^{-}\frac{\cdot}{\tilde{A}} E_{t_{i} 1} R_{t} (\frac{W_{i;t}}{W_{i;t_{i} 1}}) (\frac{G_{i;t}}{G_{i;t_{i} 1}})^{\tilde{A}} (\frac{I_{i;t}}{I_{i;t_{i} 1}})^{\cdot i 1} :$$
 (4)

where  $R_t$  is the return on the riskless or the risky asset. The intertemporal rate of substitution for labor supply can then be obtained from the following log linearization of (5)

$$\text{(Inl}_{i:t} = \hat{A}_1 + \hat{A}_2 \ln R_t + \hat{A}_3 \text{(Inq}_{i:t} + \hat{A}_4 \text{(Inw}_{i:t} + u_{i:t})$$

The intertemporal rate of substitution for labor supply in this regression is de ...ned as j  $\hat{A}_4$ : It can be easily veri...ed that this quantity can be related to our

<sup>&</sup>lt;sup>7</sup>0 ne can estimate and test a linear estimating equation without any extra assumptions, by combining the ...rst order conditions for consumption and leisure, and neglecting the intertemporal optimality condition. This type of analysis is referred to as estimation of the intratemporal optimality condition (e.g. seell aCurdy (1983), I Itanji (1984) and EHS (1988)).

<sup>&</sup>lt;sup>8</sup> There are a few papers in the panel data literature that provide an analysis of nonlinear Euler equations. If otz, Kydland and Sedlacek (1988) analyze a translog utility function. B ecause this utility speci...cation is quite dia erent from the ones traditionally analyzed in the literature, it is not straightforward to compare their results with other studies. If Itug and Illier (1990) analyze a utility function that contains several nonseparabilities between consumption and the leisure of dia erent members of the household. If owever, their estimates are obtained under the hypothesis of market completeness. If oreover, the econometric framework they use results in some instances in fairly imprecise estimates. The results in this paper therefore dia er quite strongly from the ones in these two studies.

parameterization in (1) because  $\hat{A}_4 = 1 = (\cdot \ \ i \ 1)$ : Whereas our analysis focuses on diagrant implications of the theory, we can therefore derive the intertemporal rate of substitution because our estimates identify preferences, if all the implications of the theory are correct. It may be that by focusing on implications of the theory that contain less interesting information on leisure choice, we end up with less precise estimates of the intertemporal rate of substitution. If owever, I want to emphasize again that an argument in favor of the analysis in this paper is that the ... rst order conditions with respect to consumption are more likely to hold than those with respect to leisure. So from the perspectice of consistent parameter estimation a comparison with parameter estimates from the labor supply literature can be very instructive.

A ...nal di¢ culty in comparing parameters estimated in this paper with those obtained in the labor supply literature is the testing method used in those studies. Whereas the labor supply literature is interested in the parameter that determines the intertemporal rate of substitution for labor supply in order to comment on its policy implications, the focus of those studies is primarily on tests of the underlying theory that agents trace or utility over time. Whereas the theory can be tested by focusing on overidentifying restrictions, often tests are executed by inspecting the signi...cance of extra coe¢ cients that are inserted into the Euler equation (seelf am (1986)). Presumably such tests are more powerful than tests of overidentifying conditions. With this approach, behavioral parameters are estimated consistently under the null that the theory is correct. However, when using this technique, rejection is the rule rather than the exception. The relevant question is therefore how to interpret the parameters if the extra variables enter signi...cantly and the theory is rejected.

The same caveat applies to the comparison between the estimates in this paper and the estimates of preference parameters obtained by several studies in the consumption literature. The object of interest in this literature is the intertemporal rate of substitution for consumption, which is usually obtained in panel data studies by estimating a log linearization of the Euler equations (3) and (4). This gives

$$\ln G_{:t} = \prod_{1} + \prod_{2} \ln R_{t} + \prod_{3} \ln I_{i:t} + u_{i:t}$$
 (8)

where the parameter!  $_2$  is the intertemporal rate of substitution for consumption, which can be related to the preference speci...cation (1) as!  $_2 = 1 = (1; \tilde{A})$ :

If the theory is rejected by the presence of extra variables in the Euler equation, the resulting estimates of preference parameters are uninterpretable. So is the estimate from the Euler equation without the extra variables, because we know that the orthogonality conditions used in estimation are false. The temptation is to always interpret estimates from studies that simply test orthogonality conditions as valuable. If onever, using a similar argument, if the orthogonality conditions are rejected, estimates of preference parameters are no more valuable than those obtained in the studies that insert extra variables in the Euler equation. The diffulty with maintaining this argument is of course that given that the implications of the theory are mostly rejected, it would suggest that we know nothing about the rate of intertemporal substitution. It seems then that implicitly, many economists accept the legitimacy of available parameter estimates, even if they are obtained using orthogonality conditions that are statistically rejected.

If overer, instead of testing the overidentifying conditions, Euler equations are also often tested in this literature with extra variables included in the speci...cation, to improve the power of the tests. The panel data literature on consumption and savings decisions surers from another problem, because many studies estimate models that are additively separable between consumption and leisure 10 It is dear from (7) that if the separability assmption is incorrect, the estimate of 1,2 will be biased. D ynamic studies of demand systems (see B rown. ing Deaton and Irish (1985)) do not make such assumptions, but they often focus on more detailed components of consumption. The work in this paper is of interest to the literature on consumption and savings because a statistically signi...cant estimate of the parameter would call into question some of its test results. To give an example, the life-cycle model is often questioned because several studies have documented a relationship between consumption patterns and variables such as age. This fact is often illustrated using regressions or a straightforward graphical analysis. If ovever, if does people compensate for lover consumption by enjoying more leisure this rejection of the life cycle model would be called into question. Without knowledge of the relative weights placed an consumption and leisure it is hard to appreciate the importance of this argument. If leisure enters the utility function signi... cantly, this is likely to intuence the point estimate of the rate of intertemporal substitution in consumption.

### 3 Data

The empirical investigation uses data from the Panel Study of Income Dynamics (PSID) for the period 1974-1987. The dataset is the same as the one used in Jacobs (1998b, 1999) and is described in the Alppendix. The PSID does not contain a satisfactory measure of total consumption. Therefore, I follow existing studies that use the PSID by using household food consumption as the consumption measure. It I ternatively are could use the Consumer Expenditure Survey (CEX) to investigate the preference speci...cations in (1) and (7). Unlike the PSID, the CEX contains data on total household consumption, and the fre quency of the data is quarterly, therefore yielding more time series observations than the PSID. If owever, in the CEX data on leisure are only available on a yearly basis. Ill creaver, the CEX is not a genuine panel dataset because house holds are only followed for ... ve consecutive time periods. O ne therefore has to resort to the construction of synthetic cohorts, and to investigate the impact of apprepation problems this approach is not necessarily appealing. In contrast, the PSID is a genuine panel dataset, even though not observations on a given household are not available in any given year.

<sup>&</sup>lt;sup>10</sup>Table 5.1 in B rowning and L usardi (199 6) presents an overview of available panel data studies in the consumption and savings literature. This table also indicates which studies control for leisure when analyzing the problem at hand. It can be seen from this table that there are not too many studies that do M creover, most of these follow the approach of N to tanasio and W eber (1995), where variables are inserted to control for leisure without formally modeling leisure choice. As a result estimates in these studies are also hard to relate to the ones obtained in this paper.

The PSID allows the construction of samples of households who are at interior solutions with respect to asset choice. This allows us to assess the quantitative importance of including households at corners in the sample. To identify households at interior solutions, I use a 1984 question from the PSID which asks households for their holdings of liquid assets and stocks. This question is the same as that used by II ankiw and I eldes (1991) and Jacobs (1998b, 1999). It is discussed in more detail in the II ppendix. The question allows the construction of ... ve successively smaller samples: a sample inducing all households that satisfy the selection or iteria, a sample inducing only households who ful... If the selection or iteria and who have nonzero holdings of the relevant asset, and samples inducing only households who ful... II the selection or iteria and who have holdings of the relevant asset larger than \$1,000, \$10,000, and \$100,000 respectively. Unfortunately the two smallest samples are too small to yield reliable econometric results. Parameter estimates are imprecise and not very robust R esults are therefore only reported using the ... rst three samples.

Returns an stock and bondmarkets are constructed as follows yearly returns are computed as the average of twelve returns on one year investments which expire at the end of every month of the year. This construction is motivated by the interpretation of consumption as yearly totals (‡ows) and not stocks at one point in time. The bond returns are returns on rolling over three month treasury bills and are obtained from III coodys. The stock returns are returns on the Standard and Poor's 500 composite.

# 4 Estimation and Testing

A swith most other available panel datasets, the household and not the individual is the unit of observation in the PSID. This complicates empirical testing because preferences are de...ned at the level of the individual. To resolve this issue, I include an exponential function of household size in periods that and the Euler equation. A nother issue of interest is that many available panel data studies, especially those in the labor economics and consumption literature, estimate preferences conditional on demographic variables such as age and manital status and some of the agent's decision variables such as education. For purposes of comparison with these earlier studies, I therefore estimate the Euler equations with and without an exponential function of such variables included as preference shifters. The most general speci... cation of the Euler equations is therefore given by

$$1 = {^{-}E_{t_{i}}}_{1}R_{t}(\frac{G_{;t_{i}}}{G_{;t_{i}}})^{\bar{A}_{i}}(\frac{I_{i;t}}{I_{i;t_{i}}})^{-} \exp[\%_{1} fam_{i;t} + \%_{2} fam_{i;t_{i}})] \exp[\%_{1} demq_{;t}]. \tag{9}$$

where  $fam_{i;t}$  stands for family size in period t,  $demq_{;t}$  stands for a vector of preferenceshifters at time t,  $\%_1$  and  $\%_2$  are scalar parameters and % is a vector of parameters. A smentioned before, many studies in the consumption and labor

economics literature use panel data to investigate linearizations of (8). The most general linearization investigated in this paper is

B y estimating the parameters in (1), we can retrieve the parameters  $\tilde{A}$  and that characterize the utility function (1).<sup>11</sup>

$$\left[\frac{1}{T} \underset{t-1}{\overset{\mathbf{X}}{\times}} \nabla_{t}(A)^{G}\right] W \left[\frac{1}{T} \underset{t-1}{\overset{\mathbf{X}}{\times}} \nabla_{t}(A)\right]$$
(11)

for an arbitrary III xIII weighting matrix IV . From II ansen (1982) we know that

$$J_{T} = T \left[ \frac{1}{T} \sum_{t=1}^{\mathbf{X}} \overline{\mathbf{t}} \mathbf{p}(A)^{2} \right] \mathbf{p} \left[ \frac{1}{T} \sum_{t=1}^{\mathbf{X}} \overline{\mathbf{t}} \mathbf{p}(A) \right]$$
(12)

is distributed  $\hat{\mathbb{A}}^2_{\mathbb{H}^-_{i}K}$ , where  $(b=pbpc)_i pb=1_i$   $(pqw_i p)_i pqw_i$ , and  $(pqw_i p)_i pqw_i$ , and  $(pqw_i pqw_i pqw_i)_i pqw_i$  is the generalized inverse of 0. A II matrices in (11) are evaluated at  $(pqw_i pqw_i pqw_i pqw_i)_i pqw_i$ 

<sup>11</sup>H ansen and Singleton (1983) use time series data and a lognormality argument to estimate a linearized version of the Euler equation in structural as opposed to reduced form. This allows them to identify additional preference parameters such as the discount factor. In this paper we follow the papers in the panel data literature and estimate the reduced form (9) to illustrate the discrete equation (9) and the linearized equation (9).

of the parameters that minimizes (10). The covariance matrix  $^{f b}$  is computed as

$$\mathbf{b} = \frac{1}{T} \mathbf{X}_{t=1} \left[ \mathbf{\overline{t}}_{\mathbf{\overline{t}}_{1}} \right] \left( \frac{1}{T} \mathbf{X}_{t=1} \mathbf{\overline{t}}_{\mathbf{\overline{t}}_{2}} \right) \left[ \mathbf{\overline{t}}_{\mathbf{\overline{t}}_{1}} \right] \left( \frac{1}{T} \mathbf{X}_{t=1} \mathbf{\overline{t}}_{\mathbf{\overline{t}}_{2}} \right) \right]^{O}. \tag{13}$$

The covariance matrix of T<sup>1=2</sup> (Á; 🏲) can be computed as

$$(PQW | PD)^{i} (PQW | PW | PD) (PQW | PD)^{O_{i}}$$
 (14)

The small-sample reliability of parameter estimates and test statistics depends on the weighting matrix  $\mathbb{W}$ . Define for appiven 1 xML vector of instruments  $I_t = (\frac{1}{H_t}) (z_{1;t_1}^l, \ldots, z_{1;t_1}^M)$ . The inverse of  $I_{t=1}^T I_t^Q I_t$  is used as the weighting matrix  $\mathbb{W}$ . This choice of  $\mathbb{W}$  exectively reduces the minimization in (10) to  $\mathbb{W}$  L 2SL S.

Estimation and test results are reported for ...ve di¤ erent instrument sets. Family size in periods t and till 1 is included in every instrument set. The ...rst instrument set used contains seven instruments: besides the family variables, it contains a constant, the riskless rate of return lapped once, the riskly rate of return lapped once, the occupational unemployment rate lapped once and the occupational unemployment rate lapped once interacted with age and education of the household head. The second instrument set contains deven instruments: besides the family variables there is a constant, the riskless rate of return lapped once and twice, the risky rate of return lapped once and twice, the occupational unemployment rate lapped once interacted with age and education of the household head, the occupational unemployment rate lapped twice interacted with age and education of the household head, the industry unemplayment rate lapped once interacted with age and education of the household head and the industry unemployment rate lagged twice interacted with age and education of the household head. The third instrument set contains nine instruments: it includes family size in periods t and ti, 1, a constant, the riskless and risky rates of return lapped once, the occupational unemployment rate lapped once, the occupational unemployment rate lapped once interacted with age and education of the household head, the industry unemployment rate lapped once and the industry unemployment rate lapped once interacted with age and education of the household head. The fourth and ... fth instrument sets are used to investigate the robustness of estimation and test results w.r.t. the indusion of preference shifters. B oth include deven instruments the nine instruments inducted in instrument set three, and the two demographic variables inducted in the Euler equation. In instrument set four these demographic variables are age and age squared. In instrument set ...ve the demographic variables are education and education squared.

### 5 Empirical Results

Tables 1 through 9 present results obtained using the ... ve di¤ erent instrument sets. Each table contains two parts, with the ... rst part presenting estimation and test results for the riskless asset and the second part presenting estimation and test results for the riskless asset and the second part presenting estimation and test results for the riskly asset 1 ... rst clasuss the results in tables 1 through 3, which contain estimation and test results using instrument sets one, two and three. Subsequently 1 clasuss the results in tables 4 through 9, which analyze a number of robustness issues, such as the speci... cation of leisure time, the importance of linearization and the importance of preference shifters in the Euler equation.

Tables 1, 2 and 3 present estimation and test results for instrument sets one two and three. Instrument set one in table 1 contains 7 instruments, instrument set two in table 2 contains 11 instruments, and instrument set three in table 3 contains 9 instruments. Each table presents results for three di¤ erent samples. the ... rst column presents results for a ... rst sample, which contains all households who ful... If the selection or iteria speci... ed in the lappendix. This yields a sample with H = 3555 households. It must be noted that the panel is unbalanced and that the number of doservations we actually end up with in the sample is N = 18813. The second adumn in every table presents results for smaller samples: they include those households present in the ... rst sample who have nanzero hadings of the riskless asset and nanzero hadings of the risky asset. This sample contains H = 740 households and N = 5029 doservations. The third adumn contains only those households who have holdings of the riskless and the risky asset larger than \$1000, H = 413 households and N = 2990observations respectively. For each sample we present parameter estimates and standard errors for the parameters based on the utility speci...cation (1), as well as the implied point estimates of the parameters in utility speci...cation (7) that can be derived by using  $\mu = A + \cdot$  and  $\alpha = A = \mu$ .

A ...rst important observation in table 1A is that the results in columns 2 and 3 are remarkably similar. The estimates of the discount rate <sup>-</sup> are 0.824 and 0.823, respectively, which are intuitively plausible magnitudes. For the aptimization problem under study to be well de...ned, the estimate of  $\tilde{A}_i$  1 has to be recative. This is the case in both samples and the estimates are not very dimerent. If ote that we do not investigate the consumer's decision with respect to leisure, and therefore we do not need the estimate of to imply concavity in order for these estimates to make sense. If ovever, the estimates of of -1.363 and -1.602 would guarantee that we were dealing with a concave problem, if we were to analyze leisure choice. If ote also that the estimates of  $\frac{1}{2}$  and  $\frac{1}{2}$ , indicating the importance of family size, have the expected sign. Finally, the signi...cance level for the J statistics indicates that the statistical evidence against the theory is not very strong. Finally, each adumn lists the implication of the estimates of  $\tilde{A}$  and  $\cdot$  for the parameters  $^{\circ}$  and  $\mu$  in utility speci...cation (7). We obtain estimates of of 0.353 and 0.364 in columns 2 and 3 respectively and estimates of  $\mu$  of -2.109 and -2.520 respectively. The estimate for "is very dose to the calibrated 1/3 ... rst proposed by Kydland and P rescott and commonly used in the real business cycle literature. If ovever, the value for  $\mu$ , which determines the rate of relative risk aversion, is very dimerent from the parameter values used in many simulation studies.

If corner solutions are important, ane would expect the estimates in column 1 todizer from those in columns 2 and 3. A ... rst observation is that the discount rate in cdumn 1 is signi...cantly lover than the estimates reported in cdumns 2 and 3. If overer, most importantly, whereas the estimate of A is also quite dimerent, the sign of is dimerent from that of the estimates in columns 2 and 3, even though the parameter is fairly precisely estimated. It must be emphasized that this is not necessarily a problem, because we are not analyzing leisure choice, but if we obtained a similar estimate from the ... rst order condition for leisure it would indicate that our optimization problem was not well de...ned. It is perhaps more intuitively appealing to investigate the implications of these parameter estimates for the values of the parameters ° and µ in the utility speci...cation (7). We see that the weight placed on consumption is 0.097, much smaller than in columns 2 and 3. Ill creover, the parameter µ, which determines the rate of relative risk aversion, has the wrong sign. In summary, the parameter estimates for the large sample seem much less plausible that those for the smaller samples, which is reinforced by the observation that the parameters 1/4 and 1/4 also have the wrong sign. A ... rad important deservation is that the test statistic indicates little evidence against the theoretical speci...cation. This is problematic, because we do not expect the theory to hold for households who are at corners and the parameter estimates indicate the importance of these corner solutions. It must therefore be concluded that the interpretation of these test statistics is likely to be problematic in this context.

Table 1B presents estimation and test results for the Euler equation associated with the risky asset, dotained using instrument set 1. The most important doservation is that estimation and test results largely con...rm the results re ported in table 14. Parameter estimates in columns 2 and 3 are dramatically dix event from those in adumn 1. O noe appin, the parameter estimate for in column 1 would place us in the nonconcave region of the parameter space if we were to analyze the optimal leisure decision. Whereas parameter estimates for 1/4<sub>1</sub> and 1/4<sub>2</sub> have the expected sign in adumns 2 and 3, they do not in adumn In terms of the implied point estimates for parameters ° and μ in utility speci...cation (7), the estimates in column 1 imply a negative weight on leisure (° = i 1:127) and again the parameter  $\mu$  has the wrong sign. When comparing the estimates of and in columns 2 and 3 with those in table 14, it must be noted that the weight on consumption is higher and that risk aversion is higher too Finally, test statistics in columns 2 and 3 indicate more evidence against the model as compared to those in table 1%, but the test statistic in column 1 of table 11 is again surprisingly low

The results in tables 2 and 3 investigate the robustness of the results in table 1 wr.t. the choice of the instrument set. It is dear that most of the results obtained in table 1 are very robust. First, estimates for  $\tilde{A}$  and  $\tilde{A}$  in tables 2 and 3 situate us in the concave region of the parameter space, even if we were analyzing lesure choice. The implied values of and  $\tilde{A}$  and  $\tilde{A}$  are fairly robust.

across estimation exercises, even though it must be noted that the values for "dotained in table 2Å" (0.174 and 0.137) are much lower than those dotained in the other tables. Finally, in all tables the estimates of  $\cdot$  in cdumn 1 are implausible, leading to implausible values of "and  $\mu$ . Whereas the estimates of  $\frac{1}{4}$ 1 and  $\frac{1}{4}$ 2 have the expected sign in all instances in cdumns 2 and 3, this is never the case in cdumn 1. One aspect of the results that is not necessarily robust is the signi...cance levels of the test statistics. Whereas we obtain statistical nonrejections in many cases, there are several exceptions. Also, the ...noting from table 1 that test statistics are lower in cdumn 1 is not always con...med. These results are indicative of the results for the large number of instrument sets that I have investigated and that are not reported parameter estimates are very robust, test statistics are not

It is instructive at this point to relate these parameter estimates and test statistics to the results for the time separable constant relative risk aversion (TS-CRRA) speci...cation studied in Jacobs (1999). There are some interesting parallels. First, the dixerences between the parameter estimates obtained from the Euler equation for the riskless and the risky asset are in a certain sense similar to the cross in Jacobs (1999). For example, the estimates of the discount rate of dotained from the Euler equation for the risky asset in Jacobs (1999) are also smaller than those obtained from the Euler equation for the riskless asset Second, if one interprets 1; µ as the rate of relative risk aversion for the composite good as in Kydland and Prescott (1982), then the ...nding that the rate of risk aversion is larger when estimating the Euler equation for the risky asset also dotains in Jacobs (1999). If owever, there are marked di¤ erences when estimating nonseparable preferences. First, the discount rate - estimated in this study is almost always lower than the one estimated in Jacobs (1999). Second, the test statistics in this study are markedly lower than those in Jacobs (1999). It must of course be remembered that we know very little about the performance of these test statistics in small samples in a panel data context. If ovever, systematic overrejections or underrejections are equally likely to intuence the results in a similar fashion in Jacobs (1999). The results therefore suppost that leisure is an important determinant in the utility function, a condusion which is of course reinforced by the ...noting that the parameter · is usually signi...cantly estimated in adumns 2 and 3 of the tables. Third, estimation and test results in Jacobs (1999) do not vary markedly across di¤erent columns, seemingly implying that corner solutions are not very important. When taking the presence of leisure in the utility function into account, the results in tables 1 through 3 dearly show that corner solutions are critically important and cause signi...cant biases in estimated parameters.

## 6 Robustness

Inspection of tables 1 through 3 revealed that estimation results are robust with respect to the choice of instrument set. In this section we investigate robustness

of estimation and test results in other dimensions. The unifying theme of this section is that the results are not robust to a number of important assumptions.

The ...rst issue we investigate is the speci...cation of leisure time. There is a certain arbitrariness that enters the setup, because we have to construct the leisure time that enters the utility function from available data on hours worked. The results in tables 1 through 3 are obtained by specifying total time available to the consumer in the year as 58.40 hours (1.6 hours per day multiplied by 365). Leisure time is then computed as 5840 hours minus hours of work reported for that year. This construction follows the work of Ell S (1988) and II RS (1985). However, one can also justify including sleeping time in the available leisure time, or specifying that an individual has less than 16 hours available every day to divide between consumption and leisure. To investigate the dependence of parameter estimates and test statistics on this construction, table 4 presents results obtained using the second instrument set, where the total time endowment for a year is set at 8760 hours (24 hours x 365 days). When comparing the results to table 2, it is dear that the main impact of this change is that the estimated absolute value of the parameter. is larger in columns 2 and 3. A saconsequence the implied point estimates of "and \u03a4 in the utility speci...cation (7) are dimerent from the ones in table 2: ° is smaller and the rate of risk aversion µ is much larger. There is a second and less dovious di¤erence which is nevertheless interesting. The impact of the change in the speci...cation of leisure time on the results in adumn 1 is larger than on the results in adumns 2 and 3. For the results in table 41 (the riskless asset), not only is the estimate. of very dimerent from the one in table 21, but also the estimate of A. III creaver, the test statistic is much higher. This suggests that the econometric analysis of the results in column 1 is perhaps less robust than the one for the samples that contain only households at interior conditions.

Table 5 investigates howestimation and test results are a ected by estimate ing linearizations of the original nonlinear Euler equations. Strictly speaking this is not an investigation of the robustness of the results listed in tables 1 through 3, because we do not relax any of the assumptions used to obtain those results; on the contrary, to analyze the linearizations one has to make additional assumptions. A limited analysis of these linearized equations is nevertheless of interest for several reasons. First, many available studies in the consumption, savings and labor economics literature that use panel data analyze linearized Euler equations.<sup>12</sup> Il Iso, to the best of my knowledge, analysis of linearized Euler equations has only been performed for the Euler equation associated with the riskless asset 6 iven that estimation and test results in tables 1, 2 and 3 direction direction in the risks and the risks asset, it is worth investigating if this is also the case when analyzing these linearized equations. Finally, it is well known that measurement error problems can be solved in a linear context by an appropriate choice of instruments. When estimating nonlinear Euler equations, dealing with measurement error problems is much more problematic. To

 $<sup>^{12}</sup>$  It is well known that estimation and test results for these analyses can di¤ er dramatically dependent on the sample under investigation, and therefore it is di¢ oult to compare the results in tables 1 through 3 with existing parameter estimates.

some extent, analysis of the linearized Euler equations can therefore be interpreted as an analysis of robustness of the nonlinear analysis to measurement error problems.  $^{13}$ 

Table 51 presents estimation and test results for the linearization of the Euler equation (1) associated with the riskless asset, obtained using instrument set three. Comparing the results to those in table 31, it is dear that they are dramatically dimerent. First, the test statistics indicate dramatically stronger evidence against the model. Second, when using the estimates of  $!_2$  and  $!_3$  to derive implied parameters A and . , it is dear that the absolute value of both parameters is much larger in columns 2 and 3 in table 54 compared to table 31. Translating these parameters to the more intuitively appealing parameters ° and  $\mu$  of utility speci...cation (7), it is seen that the share of consumption in the utility function is roughly comparable to the one estimated in table 31. However, the parameter µ, which indicates a higher rate of relative risk aversion is dramatically higher. A third interesting observation is that the implied estimates of the parameters ° and µ in odumn 1 are much more similar to those in columns 2 and 3 than is the case in table 31. A lso, the estimates for 11 and  $\frac{1}{2}$  in adumn 1 have the expected sign, just as in adumns 2 and 3. Table 5B provides a similar analysis for the risky asset. The same observations as in table 5A apply. Interestingly, it is also the case that the estimates for the parameter of relative risk aversion are much larger than those obtained in table 51, just as the estimates dotained in table 38 are higher than those dotained in table 31.

A ...nal important robustness analysis is performed with respect to the indusion of demographics in the Euler equation. If ost of the literature on consumption, savings and labor economics estimates (linearizations of) Euler equations or demand systems by conditioning on a number of demographic variables (preferenceshifters) such as age, education, marital status, race and others (see table 5.1. in B rowning and Lusardi (199 Å). If otivation for such analyses seems dovious, for instance by plotting consumption over the life cycle as a function of age. If ovever, once one considers nonseparable preferences it is deer that such techniques may be misleading because other components of the utility function may change too. For instance, as people get older they may consume less but still con...rm to intertemporal optimization theory because they enjoy more leisure. Exactly how much more leisure they have to enjoy to make the theory work is an open question, because it determines on the parameterization of the utility function, which should be determined from the data in the ... rst place. It could

<sup>13</sup> It is dear though that this interpretation is potentially troublesome if one can investigate the robustness to measurement error by investigating linearizations, there would be no motivation for investigating the nonlinear Euler equations in the ... rst place. Just as there are costs to analyzing the nonlinear equations (higher probability of measurement error problems), there are costs to analyzing the linearized equations (is the linearization valid?). If creover, there is no evidence indicating that linearization solves all potential measurement error problems in this context, nor that measurement error strongly biases parameter estimates for this particular estimation problem. The only thing we know for sure is that measurement error problems are potentially serious in these datasets (see I Itanji (1986) and I Itanji and Siow (1987)) and that the erects of these errors are much more likely to cause problems in a nonlinear environment.

then be the case that age enters signi...cantly in the utility function because it provies for leisure, with which it is highly correlated. Therefore, his paper takes adia erent approach from much of the literature by...rst investigating the theory without preference shifters included, and then investigating the robustness of the results to the inclusion of demographics. Our results show that this may be a worthwhile approach, as the inclusion of demographics can seriously bias estimation and test results.

Table 6 presents results for the Euler equations augmented with two regressars the ace of the household head and the ace of the household head squared. Table 7 augments the Euler equations with the education of the household head and the education of the household head squared. In both cases the instrument set used is instrument set three augmented with both regressors; these instrument sets are referred to as instrument sets four and ...ve, respectively. Table & presents results using instrument set four for the Euler equation associated with the riskless asset. Comparing the results to table 31, it is dear that the point estimates of Å and · are very dix erent. Translating this into the parameters ° and  $\mu$ , it is seen that the share of leisure is very small in column 2 and negative in cdumn 3. The estimates in cdumn 1 are not very diagrent from the ones in table 31, but they were not intuitively plausible to start with. Why do these dixerences accur? A serious problem with the estimates in column 2 and 3 is that the parameter · is estimated very imprecisely. Further inspection shows that d and d are also very imprecisely estimated. A Iso, the parameters 1/4 and ½ are more imprecisely estimated than in table 31 and they dimensional between adumns 2 and 3 than is the case in table 34. It can therefore be arousd that the point estimates for A and and the resulting point estimates for and µ have to be interpreted with caution. It is tempting to conduct that a likely reason for these imprecise estimates is that age and leisure are highly correlated. If owever, comparison of tables 43 and 38 shows that the problem is more complex when analyzing the Euler equation for the risky asset, including age and age squared in the Euler equation does not signi...cantly bias the estimates of A and . This is interesting because just as in table &, the parameters of, of, 1/41 and ½2 are not very precisely estimated.

Inspection of table 7 nevertheless shows that the properties of the age variable are part of the problem. When inducing education and education squared in the Euler equation, point estimates of Aland, and resulting point estimates of and place are not dramatically arected. This ... noting can partially be explained by the fact that the parameter is estimated with similar standard errors as in table 3. If evertheless, it must once again be noticed that the parameters of and of are very imprecisely estimated and that the J statistics in table 7 are in many cases higher than the corresponding ones in table 3.

The...nal question addressed is whether the problems with estimating utility parameters in the presence of preference shifters are speci...c to the analysis of nonlinear Euler equations. A fiterall, almost all available studies estimate prefer-

<sup>&</sup>lt;sup>14</sup>T his ...ndings is not due to the indusion of too many regressors. Identical results dotain when only age and not age squared is induded in the regression.

ence parameters in the presence of preference shifters from linearized equations. B ecause these studies typically report estimates in the presence of di¤ erent sets of preference shifters, but not while excluding preference shifters, it is di¢ oult to verify from those studies whether they also incur similar di¢ culties. Tables 8 and 9 investigate this issue by induding age and age squared and education and education squared respectively in the linearized Euler equations. To ensure that the results are comparable with those for the nonlinear Euler equations, the instrument sets used are the same, namely instrument set three augmented with the regressors. These instrument sets are referred to as instrument sets 4 and 5 respectively. Table 84 shows that just as in the nonlinear case, problems occur when including the age variables in the Euler equation associated with the riskless asset. The parameter ! 3 is very imprecisely estimated and the re sulting point estimates for are not intuitively plausible. Table 8B shows that problems also obtain when analyzing the Euler equation for the risky asset, but they are less serious, just as in the nonlinear case. Finally, table 9 shows that the problems are much less serious when induding the education variables in the Euler equation as preference shifters.

## 7 Comparison With Existing Literature

This section presents a discussion of the relevance of the parameter estimates obtained in sections 5 and 6 for the existing literature. First, the implications for dynamic macroeconomics are discussed, with particular attention to those studies that employ the methodogy in Kydland and P rescott (1982), and use parameter estimates as inputs into the model. It is dear that all of the estimates of the parameter  $\mu$  in odumns 2 and 3 of tables 1,2 and 3 fall outside the range of parameter values used by Kydland and P rescott (-1, -0.5 and -0.1). It may be argued that the dia erences with the parameter range obtained in tables 1, 2 and 3 (from -1.980 to -4.515) are insigni...cant. Indeed, when interpreting 1;  $\mu$  as the rate of relative risk aversion, it is dear that the dia erences between the parameter estimates in this paper and some of the parameters estimated or proposed in the representative agent literature are small. If owever, by the very nature of these models and the methodogy used, it is often not dear howeven modest changes in parameter values in turned the results.

If any of the studies in the "real business cycle" literature and in dynamic macroeconomics in general use the preference speci...cation (7) of Kydland and P rescott (1982) and several studies use similar parameter estimates. For instance, B ackus, Kehoe and Kydland (1992) use  $\mu=\frac{1}{1}$ . It nother potential problem for many studies in this literature is that several authors use a logarithmic preference speci...cation, which maintains achilitive separability between consumption and leisure (e.g. see Christiano and Eichenbaum (1992)). It is dear from inspecting the standard error on in tables 1 through 3 that the data conot support this hypothesis. If ore precisely, the logarithmic utility function used in those studies can be formulated as a special case of (7) with  $\mu=0$ : This hypothesis cannot be tested using the tables because we only report on

the point estimates of ° and  $\mu$  that are implied by the preference speci...cation (1). If overver, for a limited number of speci...cations we repeated the estimation using (7). Whereas ° is sometimes imprecisely estimated, estimates for  $\mu$  are always highly signi...cant 0 ur...nolings therefore indicate that the logarithmic speci...cation used in several studies is not supported by the data

The parameter estimates and test statistics reported in this paper have important implications for for the asset pricing literature, yet they do not allow for a de..nitive condusion on this topic. Following the work of Lucas (1978) and B readen (1979), a voluminous literature has investigated the empirical performance of consumption based asset pricing models, ...rst using a TS-CRRA speci...cation (see H ansen and Singleton (1982,1983,1984), M ehra and Prescott (1985) and G rossman, M elino and Shiller (1987)), and later using more general time nonseparable speci...cations (see II bel (1990), Campbell and Cochrane (1995), Constantinides (1990), Heaton (1995)). The implication of our estimates for these studies, which all (implicitly) assume additive separability between consumption and leisure, seems to be that since the parameter · is signi...cantly di¤ erent from zero, test results in those papers are severely biased. This condusion is reinforced by the ... noting that the test statistics in this paper are lower than the corresponding ones in Jacobs (1999). However, whereas the representative agent literature attaches great importance to this type of test statistics (see II arsen and Singleton (1982) and Epstein and I in (1991)), recently attention has been focused on one particular dimension of the data the so-called "unconditional restrictions" of the model. This dimension of the model is dozely associated with the so called asset pricing puzzles, such as the equity premium puzzle and the riskfree rate puzzle<sup>15</sup> The question is there fore to what extent the statistically signi...cant presence of leisure in the utility function translates into dimerent properties of the model in the dimension that matters for asset priging puzzles.

This question cannot be readily answered using the test results in this paper. 16 However, at the very least, the statistically signi... cant and intuitively

<sup>&</sup>lt;sup>15</sup>B ecause to a certain extent this terminology is a misnomer, it is instructive to indicate exactly what is meant with unconditional information. In a 6 M M setup such as the one in this paper, this information can be thought of as exploiting the orthogonality between the Euler equation error and a constant. In contrast, the estimation and testing in this paper exploits orthogonality between the Euler equation and a constant as well as other variables in the agent's information set. To appreciate that this direction can be important, with a TS-CRRM speci...cation unconditional information yields large estimates of the rate of risk aversion, whereas conditional information such as the one used in this paper yields much smaller estimates (see H ansen and Singleton (1982, 1983) and 6 rossman, M elino and Shiller (1987)).

<sup>&</sup>lt;sup>16</sup>For nonseparable preferences such a test is less straightforward than in the separable case. In the TS-CRR in case, one can simply use the orthogonality condition associated with the riskless asset and a constant and the orthogonality condition associated with the riskly asset and a constant to determine the two parameters of interest, the discount rate and the rate of relative risk aversion. In the nonseparable case we need to determine three parameters, so we need to add an orthogonality condition involving a third asset. If one adds a third asset that has di≖erent time series properties from the riskless and risky asset, the danger is that test results will be impacted by the choice of this asset. The best solution is probably to work with the riskless asset and two risky assets with properties similar to the market return.

plausible values of suggest that nonseparable utility functions are worth looking into to solve asset pricing puzzles. A loop it was mentioned previously that the representative agent...nance literature has not focused on leisure in the utility function because aggregate leisure is too smooth to signi...cantly arect the intertemporal rate of substitution, even though it enters the utility function signi...cantly when estimating Euler equations for aggregate data (See Ell S (1988)). Unlike aggregate leisure, household leisure is very variable over time. These ...nolings, together with those of Jagannathan and W ang (1995), who show that human capital variables can play an important role in cross sectional asset pricing suggest that nonseparable utility functions are worth looking into to solve asset pricing puzzles.

A comparison with available estimates of preference parameters in the representative agent literature also shows that relaxing the assumptions necessary for appreciation may change the role of leisure in asset pricing models. III RS (1985) and EHS (1988) provide a representative agent analysis of nonlinear Eu ler equations.<sup>17</sup> However, their setup di¤ers from ours in several respects. First, both studies use information from the ... rst order condition for leisure as well as the ... rst order condition for consumption. Second, whereas II RS (1985) study time separable preferences, EHS (1988) exclusively study preferences with time nonseparabilities modeled as in Kydland and Prescott (1982). Conditional on these interpretational di¢ culties, the di¤ erences between estimates in those studies and ours are striking. When MRS (1985) exclusively use the intertemparal optimality condition and the ... rst order condition wrt consumption, they do not dotain statistically signi...cant parameter estimates. Perhaps unsurprisingly, estimation results are also not robust. This contrasts with the estimates dotained by Ell S. They report two estimates of µ; 0.85 and 0.8 respectively (their third estimate of -0.16 is obtained after adjusting for taxes). While these estimates are in the concave region of the parameter space, the implied value of the rate of relative risk aversion is much lower than the one dotained in this pa per. A Iso, EH S's estimate of the share parameter" is around 0.15, much lower than ours. Besides the dovious remark that this estimate may be strongly affected by the time nonseparabilities in leisure, it must also again be noted that this estimate is a ected by the de..nition of total available time. Using the alternative de...nition of leisure time in table 4, our estimates are much doser to those of Ell S (1988). Ell S also note (1988, p.6) that changes in their de.. nition of leisure time could bring their estimates more in line with our estimate (and

A nother possible approach, following H ansen and Jagannathan (1991), is to evaluate the two orthogonality conditions obtained by combining Euler equations (3) and (4) with a constant for direct values of  $\bar{}$ ; and  $\bar{}$ . The question is then if given the data, we can ... not solutions for the Euler equations for a priori plausible values of the behavioral parameters.

 $<sup>^{17}</sup>$ III ote again that for this discussion studies are gruped together mainly according to methodology. The studies by III R.S. (1985) and EH.S. (1988) are intimately related to and motivated by the "real business cycle" methodology pioneered by Kydland and P rescott (1982). However, because they provide tests of Euler equations they are intimately related to the asset pricing tests of Hansen and Singleton (1982) and for that reason they are discussed here. This goes to prove once again that the results in this paper are of interest in several research ...elds, but these ...elds are intimately connected.

Kydland and Prescott's (1982) calibrated value).

Finally, estimates of the parameters à and · obtained in tables 1 through 3 are of interest to the consumption and labor supply literature. The estimates are of interest to the consumption literature because of several reasons ...rst, in the consumption literature many tests are constructed using additive separability. The statistically signi...cant estimates of in tables 1 through 3 show that this assumption is not supported by the data. Second, the estimate of!  $_2 = _i 1 = (\tilde{A}_i 1)$  in (10) is of interest because it is the rate of intertemparal substitution in consumption. Because our estimates of A in columns 2 and 3 of tables 1 through 3 are between -1.347 and -2.749, our results imply a rate of intertemporal substitution of consumption between 0.266 and 0.426 It is not straightforward to compare these point estimates to the ones that are available in the literature, because a wide range of parameter estimates have been reported in existing research. A nother caveat is that most available es timates of the rate of intertemporal substitution in consumption are dotained using an additively separable speci...cation which is TS-CRRA in consumption. In this context the intertemporal rate of substitution in consumption is simply one divided by the rate of risk aversion. In the representative agent literature, most studies point towards a very large rate of risk aversion and therefore a small intertemporal elasticity (e.g. see H. all (1988), M. ehra and P. rescott (1985). and Hansen and Jagannathan (1991)). The exception is the evidence in Hansen and Singleton (1982, 1983): when using conditional information, their estimates of the intertemporal substitution elasticity are in line with ours. In the panel data li terature, under additive separabili tydi¤ erent studies have reported a wide range of point estimates. I eldes (1989) reports estimates in line with the small elasticities in the representative agent literature. The results in R unkle (1991) and Keane and R unkle (1992) are larger, but smaller than the estimates in this paper. A transio and W eber (1995), using synthetic cohorts, from the CEX ...nd small dasticities and A transio and B rowning (1995) using B ritish data...nd a wide variety of estimates. In the representative agent literature that investicates nonseparable preferences, the study by EHS (1988) is a natural reference point, because the results in III RS (1985) are not very robust. Their estimates of the intertemporal rate of easticity are actually larger than ours. A I tonji and If am (1990) present panel data evidence under nonseparabilities and also ... not small dasticities.

It is not the objective of this paper to attempt to recordile these dimerent estimates. They are determined, among other things, by dimerences in data selection and econometric methodogy. If owever, comparison of the nonlinear results in table 31 and the estimation results for the linearized equation in table 54 suggests are potential explanatory factor. many elasticities estimated in the panel data literature are estimated using linearized Euler equations, and they are far more in line with the results in table 54. Then with those in table 31. Whereas the existing literature does not investigate the Euler equation associated with the risky asset, inspection of the results in table 5B indicates intertemporal substitution elasticities even smaller than the ones reported in existing literature when using the riskless asset

If ow do the results in tables 1 through 3 reject on the labor economics literature? A soutlined previously, the intertemporal elasticity of labor supply can be computed as i = (i + 1): i iven that the estimates of i in odumns 2 and 3 of tables 1 through 3 are between -1.26 and -3.188, this implies an elasticity of substitution between 0.238 and 0.441. This range of estimates is higher than most estimates available in the labor supply literature. This is of critical importance because the lowestimates in the labor supply literature have traditionally been invoked to argue that intertemporal models did not stand a chance to explain the data. There are several potential explanations for this diagrance between our estimates and those available in the literature First, note that once again the de...nition of leisure plays a role. Table 4, with the alternative de..nition of leisure has estimates of of -5.661 and -4.487 respectively, which leads to lower elasticity estimates. However, once again, a mare striking dia erence obtains when we investigate linearizations of the Euler equations in table 5. In table 51, where we investigate the Euler equation associated with the riskless asset, the estimates of imply estimates of the substitution elasticity of approximately 0.1, more in line with estimates in the literature. In table 5B implied values of the substitution elasticity are actually much smaller than most estimates available in the literature! (Il ote again that the Euler equations associated with the risky asset are not investigated in the labor supply literature). It therefore seems that some of the dia erences between the estimates in this paper and those in the literature seem to be determined by the di¤erence in estimation methodology (linearized versus nonlinear Euler equation), and not by which ... rst order condition is being studied (consumption versus leisure).

Finally, what is the role of preference shifters? It is mentioned in Section 6 for some Euler equations the presence of preference shifters yields imprecise parameter estimates. For those that are estimated precisely, the implied estimate of the parameter and I in tables 6 and 7 are not too much a ected. The same conslusion holds for the estimates of the linearized Euler equations in tables 8 and 9. The dia erences in intertemporal elasticities with the existing literature are therefore most likely not due to the rich array of preference shifters that is estimated in many available studies.

#### 8 Candusian

This paper estimates and tests nonlinear intertemporal Euler equations using a preference speci...cation that is nonseparable in consumption and leisure. It ...nots that for samples that only contain households at interior solutions, parameter estimates are intuitively plausible. Parameter estimates indicate that the optimization problems under study are concave, which is reassuring. Further, for Cobb-D audias preferences, the implied point estimates for the share

 $<sup>^{18}\,\</sup>text{M}$  convexi...cation argument devised in R ogerson (1988) and used in H ansen (1985) can reconcile low intertemporal substitutability of individual agents with high intertemporal substitutability of the representative agent

of consumption in the utility function and the rate of relative risk aversion are intuitively appealing. For a sample that also contains households at corner solutions, parameter estimates are almost always hard to interpret. This ...not ing indicates that participation in asset markets is of critical importance when estimating preference parameters. As a result studies of representative agent models are likely to be severely biased. This ...noting contrasts with the ...notings in Jacobs (1999), who studies a TS-CRRA preference speci...cation. Whereas estimation results die endependent on whether one analyzes the Euler equation associated with the riskless on the risky asset, these die energes are relatively minor.

A critical di¤erence between the methodology used in this paper and the approach of most of the existing panel data literature is that almost all available papers investigate linearizations of the original nonlinear Euler equations. It is shown that when investigating linearizations, parameter estimates are dra matically diagrant Most importantly, the implied rate of relative risk aversion is much higher than the one obtained using nonlinear analysis. Interestingly, this ...noting matches the ...notings of Jacobs (1998) for a TS-CRRA utility function. Even though it may be tempting to conducte from this that linearization biases parameter estimates, this is not necessarily the case. A nother possible interpretation is that parameter estimates dotained from the nonlinear analysis are biased because of measurement error problems. A nother dia erence between this paper and many others is that besides controls for family size, no other pref erence shifters are included in the Euler equations. When including preference shifters, it is shown that the presence of age variables in the Euler equations leads to imprecise estimates and a lack of robustness when analyzing the Euler equation for the riskless asset.

B ecause preference speci...cation lies at the basis of most economic analysis, these...notings are of substantial interest. I discuss in detail the importance of the ...nolings in this paper for three dixerent but interconnected research areas 1) intertemporal asset pricing 2) the literature on consumption, savings and intertemporal labor supply, and 3) the "real business cycle" literature and more opnerally the intertemporal macroeconomics literature. For the asset priding literature, the ...nolings in this paper suppost that nonseparabilities may substantially a ect the performance of equilibrium asset pricing models. If owever, because of the nature of the restrictions that are investigated in this paper, estimation and test results do not give a de.. nitive answer to the potential of nonseparabilities to solve asset pricing puzzles, such as the equity premium puzzle and the risk free rate puzzle. The results are of great interest for the literature on consumption, savings and intertemporal labor supply. First and foremost, the results indicate that many test results in the consumption literature are invalid because of a separability assumption. Ill creover, when estimating the intertemporal elasticity of consumption, our estimates are much higher than most of the ones available in the literature. A similar ...nding holds for our estimates of the intertemporal elasticity of labor supply. It is shown that some of the lower estimates in the literature may obtain because of log linearization, although it must be emphasized that this paper does not or er convincing arguments toprefer are type of estimation method over the other. Finally, the high estimates of intertemporal substitution easticities are in principle good news for an extensive literature which investigates dynamic general equilibrium models by simulation. If ovever, our estimates show that the parameters calibrations used in many of these studies are subject to criticism. It is diff cult to judge to what extent these...notings invalidate the condusion reached by those models.

These ....notings suggest a number of questions and extensions. First, it is dear that the use of the PSID, which forces us to use food consumption, may be a problem. The CEX or ers data on total consumption, but is not a panel. Perhaps investigation of clatasets from countries other than the United States will be helpful here, but such analysis may not be instructive to explain phenomena that are typical to the United States, such as the low savings rate Second, some of the comparisons made with other studies in the literature are imperfect, because many existing studies use time nonseparabilities. This observation suggests an analysis of those more general utility functions. Finally the patterns of the test statistics are sometimes diffully to understand. All detailed study of the performance of the test statistics used here in a panel data context is desirable.

# 9 A ppendix: Data Selection

This appendix describes the data selection procedure for the P and Study of Income D ynamics (PSID) data used in the empirical analysis. I use data from the PSID for the years 1974 to 1987. The data is taken from the 1987 respondent and non-respondent...les of the PSID and includes all data on families headed by a male, including single males. O beer vations on individuals in the poverty subsample are included in the sample if they ful.... I the selection or iteria.

The central issue is the construction of the consumption measure. The most important problem is that the PSID allows only the construction of a measure of food consumption. I therefore follow the existing literature by using food consumption as the consumption measure. A nother problem is that this food consumption measure is desired at the household level, and the theory is at the level of an individual agent. The latter problem is solved by working with household consumption and including a function of family size in the Euler equation. A II consumption measures in the PSID are in nominal terms. They are converted to real terms by detaiting by the food consumption price index, which is obtained from the Economic Report of the P resident.

The measure of consumption is constructed by aggregating i) money spent on food in restaurants; ii) money spent on food in the home which is not purchased with food stamps; and iii) the monetary value of food obtained through food stamps. The expenditure information on food in and outside the house in interview year t-1 is interpreted as referring to year t-0 ther authors have as

sumed that expenditure on food consumed in the home and restaurants in year tis a weighted average of the responses from interview year till and interview year till usually with the respective weights being .75 and .25. It must be noted that this construction of the consumption measure interprets the relevant PSID questions as referring to a tow variable, as opposed to a stock at a point in time. The reason that dimerent studies have not treated this information in a consistent way is that the questions asked are not without ambiguity (Seell Itug and III iller (1990), It all and III ishkin (1982), III ankiwand I eldes (1991), R unkle (1991) and I eldes (1989) on this issue). S took and bond returns are constructed tomatch the construction of the consumption series.

For each Euler equation, results are reported for three dia erent samples. If ... rst sample is the same regardless of the Euler equation under investigation and includes all observations for which the following data selection or iteria are satis...ed:

- <sup>2</sup> the household head has to be between 25 and 60 years of age.
- $^{2}$  yearly hours worked by the household head have to be between 100 and 41.60.
- 2 total real food consumption in 1987 ddlars has to be less than \$12,000 per person and more than \$720 per person and total real family food consumption has to be less than \$30,000.
- 2 there can be no missing data on the demographic information used in the estimation exercises. The dimerent demographic variables used as regressors are age of the head, age of the head squared, family size in period to family size in period to the head squared, family size in period to family size in period to the head squared, family size in period to family size in period to the head squared, family size in period to family size in period to the head squared family size in period to the head is used as a selection or territor because it is used in the construction of the instrument set. It must be noted that some estimation exercises using these demographics are not reported in the paper.

This ...rst sample has 18813 doservations. For all Euler equations under investigation, estimation and test results are reported for two other samples, which are meant to include only households with strictly positive holdings of the relevant assets. The samples are created by including only households who state that they have nonzero holdings of the relevant asset, or holdings larger than \$1,000. To select these households, a series of 1984 questions from the PSID are used. These questions essentially ask households whether they have positive holdings of a relatively riskless and/or a risky asset at that time. Speci...cally, for the riskless asset the questions (questions # V 10917 through # V 10921) ask

2 "D o you (or anyone else in your family living there) have any money in drecking or savings accounts, money market funds, certi...cates of deposit, government savings bonds, or T ressury bills, including IRA 's?"

- <sup>2</sup> if at mative answer to i)
- 2 "If you added up all such accounts for all of your family living there, about how much would they amount to right now?"
- <sup>2</sup> iii) if no answer to ii)
- 2 "W ould it amount to \$10,000 or more?" and dependent on this answer
- 2 "\$1,000 ar mare?" ar "\$100,000 ar mare?" For the risky asset, questions # V 109 12 through # V 109 16 are
- 2 "D oyou (or anyone in your family living there) have any shares of stock in publicly held corporations, mutual funds, or investment trusts, including stocks in IR # s?"
- <sup>2</sup> if at mative answer to i)
- <sup>2</sup> "If you sold all that and paid or everything you owed on it, how much would you have?"
- <sup>2</sup> if no answer to ii)
- <sup>2</sup> "W auld it amount to \$10,000 or more?" and dependent on this answer
- 2 "\$1,000 armare?" ar "\$100,000 armare?"

The main purpose of using these questions is a comparison of estimation and test results between

the sample that only includes households at interior conditions and the sample that also includes households at corner solutions. The analysis of the sample that only include households with asset holdings larger than \$1,000 is interesting from two perspectives. First, this sample is less likely to contain dassi...cation errors (households at corner solutions), and therefore it is interesting to compare them to the sample that also includes households at corner solutions. It loop a comparison of estimation and test results between the dimerent samples with positive asset holdings can indicate whether they have dimerent characteristics. It must be noted that the questions listed above also allow construction of a sample of households with asset holdings larger than \$10,000 and \$100,000. If owever, these samples are not used in the analysis because they do not yield robust estimation results.

It must also be noted that the selection criterion used in this paper is potentially problematic. The most important problem is that for every year that a household is included in the sample, it is dassimed as an assetholder or a non-assetholder on the basis of this 1984 question. This may obviously give rise to misclassimications. A log a potential problem with the interpretation of the results is that the dimerence between the second and the mirst sample for the

analysis of a given Euler equation is not necessarily totally made up by house holds who are non-assetholders in 1984. A household may simply not be present in the sample in 1984, yet be an assetholder in every other year. For a more detailed discussion of the 1984 PSID question see M ankiwand I edgs (1991).

#### R eferences

- [1] A bel, A notrew B., 1990, A seet prices under habit formation and catching up with the Joneses, A merican Economic Review Papers and Proceedings 80, 38-42.
- [2] A Itanji, Joseph G., 1982, The intertemporal substitution model of labour market ‡uctuations an empirical analysis, Review of Economic Studies 49, 783-824.
- [3] A Itanji, Joseph G., 1986. Intertemporal substitution in labor supply: exidence from micro data, Journal of Political Economy 94, \$176\$215.
- [4] A Itanji, Joseph G. and A loysius Siow, 1987, Testing the response of consumption to income changes with (naisy) panel data, Q uarterly Journal of Economics 102, 29 3-328.
- [5] A. Itanji, Joseph G. and John C. H. am, 1990, Intertemporal substitution, exagencity and surprises: estimating life cycle models for Canada, Canadian Journal of Economics 23, 1-43.
- [4] A Itug SumruandR obertA . M iller, 1990, H ouseholdchaices in equilibrium, Econometrica 58, 543-570.
- [7] A ttanasio, O razio P. and M artin B rowning 1995, Consumption over the lifecyde and over the business cycle, A merican Economic Review 85, 1118-1137.
- [8] A transiq O razio P. and G uglielmo W eber, 1995, Is consumption growth consistent with intertemporal optimization? Evidence from the consumer expenditure survey, Journal of Political Economy 103, 1121-1157.
- [9] Backus, David K., Patrick J. Kehoe and Finn E. Kydland, International real business cycles, Journal of Political Economy 100, 745-775.
- [10] Basak, Suleyman, 1999, On the ‡uctuations in consumption and market returns in the presence of labor and human capital: an equilibrium analysis, Journal of Economic Dynamics and Control 23, 1029-1064.
- [11] Bodie  $\it l$  vi, Robert C. Mertan and William F. Samuelson, 1992, Labor supply  $\it t$  exibility and particliochoice in a lifecy demodel, Journal of Economic Dynamics and Control 16, 427-449.
- [12] B reeden, D auglas T., 1979, A n intertemporal asset pricing model with stochastic consumption and investing apportunities, Journal of Financial Economics 7, 265-296
- [13] B reeden, D ouglas T., M idhael G ibbons and R obert Litzenberger, 1989, Empirical tests of the consumption-based CA PM, Journal of Finance 44, 231-262.

- [14] B rowning III artin and II nnamarial usardi, 199 6 H ousehold saving micro theories and micro facts, Journal of Economic Literature 34, 179 7-1855.
- [15] B rowning III artin, II ngus S. D eaton and III argaret Irish, II pro...table approach to labor supply and commodity demands over the life cycle, Econometrica 53, 503-544
- [14] Campbell, John Y. and John Cochrane, 1995, By force of habit: A consumption-based explanation of aggregate stock market behaviour, N. B. ER. working paper no. 4995, N. ational Bureau of Economic Research, Cambridge, N. A.
- [17] Campbell, John Y., A. ndrewW. L. q. and A. Craigll adKinlay (1997), "The Econometrics of Financial M. arkets," Princeton University Press.
- [18] Christiano, Lavrence J. and Martin Eichenbaum, 1992, Current real business cycle theories and aggregate labor-market ‡ uctuations, American Economic Review 82, 430-450.
- [19] Cochrane, John, 1991, A simple test of consumption insurance, Journal of Political Economy 99, 957-976
- [20] Cochrane, John and Lars Peter Hansen, 1992, Asset pricing lessons for macroeconomics, in 0. Blanchard and S. Fischer, eds.: 1992 NB ER Nacroeconomics Annual (National Bureau of Economic Research, Cambridge, NA).
- [21] Constantinides, 6 earge M ., 1990, H abit formation: A resolution of the equity premium puzzle, Journal of P ditical Economy 98, 519-543.
- [22] Deaton, Il ngus S., 1992, Understanding consumption (0 xford University Press, Il ewYork, Il Y).
- [23] D eaton, A ngus S. and John III uellbauer, 1980, Economics and consumer behavior, II ew Y ork, Cambridge U niversity P ress.
- [24] Detemple, Jerome, 199 & Intertemporal asset pricing with incomplete markets and nontracked assets, working paper, III di ill University.
- [25] Detemple, Jerome and Fernando I apatero, 1991, Il seet prices in an exchange economy with habit formation, Econometrica 59, 1633-1657.
- [24] Eichenbaum, III artin, L ars Peter II ansen, and Kenneth Singleton, 1988, I time series analysis of representative agent models of consumption and leisure choice under uncertainty, Q uarterly Journal of Economics 103, 51-78.
- [27] Epstein, Larry G. and Stanley E. I in, 1991, Substitution, risk aversion, and the temporal behavior of asset returns: an empirical analysis, Journal of Political Economy 99, 263-287.

- [28] Fama, Eugene, 1991, E¢ cient capital markets: II, Journal of Finance 46, 1575-168.
- [29] Grossman, Sanford J., Meling Angeloand Robert J. Shiller, 1987, Estimating the continuous time consumption based asset pricing model, Journal of Business and Economics Statistics 5, 315-327.
- [30] Hall, Robert E., 1988, Intertemporal substitution in consumption, Journal of Political Economy 9 6, 339-357.
- [31] Hall, Robert E., and Frederic S. Mishkin, 1982, The sensitivity of consumption to transitory income. Estimates from panel data on households, Econometrica 50, 461-482.
- [32] Ham, John, 1986 Testing whether unemployment represents intertemporal labour supply behaviour, Review of Economic Studies 53, 559-578.
- [33] Hansen, Gary, 1985, Indivisible labor and the business cycle, Journal of Monetary Economics 16, 309-327.
- [34] Hansen, Lars Peter, 1982, Large sample properties of generalized method of moments estimators, Econometrica 50, 1029-1054.
- [35] Hansen, Lars Peter and Ravi Jagannathan, 1991, Implications of security market data for models of dynamic economies, Journal of Political Economy 99, 225-262.
- [36] Hansen, Lars Peter and Kenneth Singleton, 1982, Ceneralized instrumental variables estimation of nonlinear rational expectations models, Econometrica 50, 126-1286
- [37] Hansen, Lars Peter and Kenneth Singleton, 1983, Stochastic consumption, risk aversion and the temporal behavior of asset returns, Journal of Political Economy 91, 249-265.
- [38] If ansen, Lars Peter and Kenneth Singleton, 1984, Errata, Econometrica 52, 26-263.
- [39] Hayashi, Fumiq Joseph II Itanji and Laurence Kotliko: , 1994, Risk sharing across and within families, Econometrica 64, 261-294.
- [40] H eaton, John, 1993, T he interaction between time nonseparable preferences and time aggregation, Econometrica 61, 353-385.
- [41] If eaton, John, 1995, If in empirical investigation of asset pricing with temporally dependent preference speci...cations, Econometrica 63, 631-717.
- [42] Heaton, John and Deborah Lucas, 1996, Evaluating the exects of incomplete markets for risk sharing and asset pricing Journal of Political Economy 104, 443-487.

- [43] Hedrman, James J. and Thomas E. III a Curdy, II. life cycle model of female labour supply, Review of Economic Studies 47, 47-74.
- [44] Hotz, V. Joseph, Finn E. Kydland and Guilhermel. Sedlacek, 1988, Intertemporal preferences and labor supply, Econometrica 5,6, 335-360.
- [45] Jacobs, Kris, 1998a, Estimating the rate of relative risk aversion: interpreting the evidence from panel data, working paper, II dill University.
- [46] Jacobs, Kris, 1998b, Il seet market data do not imply high risk aversion: reassessing the evidence on asset pricing puzzles, working paper, III di ill University.
- [47] Jacobs, Kris, 1999, Incomplete markets and security prices: obasset pricing puzzles result from aggregation problems?, Journal of Finance 54, 123-164.
- [48] Jagannathan, Ravi and I henyu W ang 199 6 The conditional CA PM and the cross-section of expected returns, Journal of Finance 51, 3-53
- [49] Keene, III idhael P. and D avid E. R unkle, 1992, O n the estimation of panel data models with serial correlation when instruments are not strictly exogenous, Journal of B usiness and Economic S tatistics 10, 1-9.
- [50] Kydland, Finn and Edward Prescott, 1982, Time to build and aggregate ‡uctuations, Econometrica 50, 1345-1370.
- [51] Lucas, Robert, 1978, A seet prices in an exchange economy, Econometrica 46,1429-1445.
- [52] III ace, Barbara, 1991, Full insurance in the presence of aggregate uncertainty, Journal of Political Economy 99, 928-956
- [53] III aCurdy, Thomas E, 1981, II nempirical model of labor supply in a life cycle setting Journal of Political Economy 89, 1059-1086
- [54] III aCuroly, Thomas E., 1983, A simple scheme for estimating an intertemporal model of labor supply and consumption in the presence of taxes and uncertainty, International Economic Review 24, 265-289.
- [55] III ankiw, III. G regary and Stephen P. I eldes, 1991, The consumption of stockholders and nonstockholders, Journal of Financial Economics 29, 113-135.
- [54] III ankiw, II. G regary, Julio R otemberg and L awrence Summers, 1985, Intertemporal substitution in macroeconomics, Quanterly Journal of Economics 100, 225-252.
- [57] III ayers, D axid, 1972, III on marketable assets and capital market equilibrium under uncertainty, in: III. Jensen, ed.: Studies in the Theory of Capital IIII arkets (Præger, III ew York, III Y), 223-248.

- [58] III ayers, D axid, 1973, II onmarketable assets and the determination of capital asset prices in the absence of a riskless asset, Journal of B usiness 46, 258-276
- [59] III ehra, Rajnish and Edward C. Prescott, 1985, The equity premium: A puzzle, Journal of III one tary Economics 15, 145-161.
- [60] Rogerson, Richard, 1988, Indivisible labor, lotteries and equilibrium, Journal of III one tary Economics 21, 316
- [6] Ross, Steven, 1976, The arbitrage theory of capital asset pricing Journal of Economic Theory 13, 341-369.
- [62] R unkle, D avid E., 1991, L iquidity constraints and the permanent income hypothesis. Evidence from panel data, Journal of III anetary Economics 27, 73-98.
- [63] Sundaresan, Suresh, 1989, Intertemporally dependent preferences and the volatility of consumption and wealth, Review of Financial Studies 2, 73-88
- [64] Svensson, Lars and Ingrid Werner, 1993, Nontraded assets in incomplete markets, European Economic Review 37, 1149-1163.
- [65] Telmer, Chris I., 1993, A seet pricing puzzles and incomplete markets, Journal of Finance 48, 1803-1832
- [66] I eldes, Stephen P., 1989, Consumption and liquidity constraints, Journal of Political Economy 97, 305-346

Table 1A

Estimation and test results for the nonlinear Euler equation associated with the riskless asset. Results obtained using instrument set one.

Estimation and test results obtained by GMM estimation of the Euler equation

$$I = \boldsymbol{b} \left(\frac{c_{i,t+1}}{c_{i,t}}\right)^{\mathbf{y}-1} \left(\frac{l_{i,t+1}}{l_{i,t}}\right)^{\mathbf{k}} \left(\frac{1}{q_t}\right) \exp[\boldsymbol{p}_1 fam_{i,t+1} + \boldsymbol{p}_2 fam_{i,t}] + e_{i,t+1}$$

where  $c_{i,t}$  is consumption of household i at date t,  $l_{i,t}$  is the leisure time of household head i at date t,  $q_t$  is the (normalized) price of the riskless asset in period t,  $fs_{i,t}$  is the size of household i in period t and  $e_{i,t+1}$  is an econometric error term. Results for  $f_1$  and  $f_2$  are not reported.

	All Households	Asset Holdings > 0	Asset Holdings > 1000
b	0.676	0.824	0.823
(Standard Deviation)	(0.095)	(0.093)	(0.077)
y - 1	-0.428	-1.745	-1.602
(Standard Deviation)	(0.0.482)	(0.468)	(0.834)
k	5.298	-1.363	-1.602
(Standard Deviation)	(0.586)	(0.624)	(0.834)
$p_1$	-0.020	0.051	0.048
(Standard Deviation)	(0.032)	(0.035)	(0.029)
$p_2$	0.024	-0.038	-0.042
(Standard Deviation)	(0.027)	(0.027)	(0.024)
Number of instruments	7	7	7
Degrees of Freedom	2	2	2
J statistic	0.247	2.066	1.674
(Significance Level)	(0.883)	(0.355)	(0.433)
Implied &	0.097	0.353	0.364
Implied $oldsymbol{J}$	5.870	-2.109	-2.520
H N	3555 18813	740 5029	413 2990

Table 1B

Estimation and test results for the nonlinear Euler equation associated with the risky asset. Results obtained using instrument set one.

Estimation and test results obtained by GMM estimation of the Euler equation

$$I = \boldsymbol{b} \left(\frac{c_{i,t+1}}{c_{i,t}}\right)^{\mathbf{y}-1} \left(\frac{l_{i,t+1}}{l_{i,t}}\right)^{\mathbf{k}} \left(\frac{p_{t+1}+d_{t+1}}{p_{t}}\right) \exp[\boldsymbol{p}_{1} fam_{i,t+1} + \boldsymbol{p}_{2} fam_{i,t}] + e_{i,t+1}$$

where  $c_{i,t}$  is consumption of household i at date t,  $l_{i,t}$  is the leisure time of household head i at date t,  $p_t$  is the price of the risky asset in period t,  $d_t$  is the dividend of the risky asset in period t,  $f_{i,t}$  is the size of household i in period t and  $e_{i,t+1}$  is an econometric error term. Results for  $f_{i,t}$  and  $f_{i,t}$  are not reported.

	All Households	Asset Holdings > 0	Asset Holdings > 1000
b	0.382	0.560	0.679
(Standard Deviation)	(0.053)	(0.208)	(0.238)
y - 1	-1.176	-2.724	-2.409
(Standard Deviation)	(0.692)	(0.831)	(1.086)
k	6.672	-1.850	-2.199
(Standard Deviation)	(0.637)	(0.804)	(1.239)
$p_1$	0.010	0.010	0.062
(Standard Deviation)	(0.087)	(0.067)	(0.055)
$p_2$	0.023	-0.058	-0.050
(Standard Deviation)	(0.067)	(0.049)	(0.039)
Number of instruments	7	7	7
Degrees of Freedom	2	2	2
J statistic	0.451	6.950	6.831
(Significance Level)	(0.798)	(0.030)	(0.032)
Implied <b>£</b>	-0.027	0.482	0.390
Implied $oldsymbol{J}$	6.496	-3.575	-3.608
H N	3555 18813	740 5029	413 2990

Table 2A Estimation and test results for the nonlinear Euler equation associated with the riskless asset. Results obtained using instrument set two.

$$I = \boldsymbol{b} \left(\frac{c_{i,t+1}}{c_{i,t}}\right)^{\mathbf{y}-1} \left(\frac{l_{i,t+1}}{l_{i,t}}\right)^{\mathbf{k}} \left(\frac{1}{q_t}\right) \exp[\mathbf{p}_1 fam_{i,t+1} + \mathbf{p}_2 fam_{i,t}] + e_{i,t+1}$$

 $I = \boldsymbol{b}(\frac{c_{i,t+1}}{c_{i,t}})^{\mathbf{y}-l}(\frac{l_{i,t+1}}{l_{i,t}})^{\mathbf{k}}(\frac{1}{q_t})\exp[\boldsymbol{p}_1 \ fam_{i,t+1} + \boldsymbol{p}_2 \ fam_{i,t}] + e_{i,t+1}$  where  $c_{i,t}$  is consumption of household i at date t,  $l_{i,t}$  is the leisure time of household head i at date t,  $q_t$  is the (normalized) price of the riskless asset in period t,  $fs_{i,t}$  is the size of household i in period t and  $e_{i,t+1}$  is an econometric error term. Results for  $f_1$  and  $f_2$  are not reported.

	All Households	Asset Holdings > 0	Asset Holdings > 1000
b	0.760	0.783	0.894
(Standard Deviation)	(0.062)	(0.085)	(0.076)
y - 1	-2.770	-1.674	-1.347
(Standard Deviation)	(0.536)	(0.495)	(0.414)
k	4.624	-3.188	-2.182
(Standard Deviation)	(0.699)	(0.633)	(1.101)
$oldsymbol{p}_1$	-0.019	0.043	0.033
(Standard Deviation)	(0.029)	(0.030)	(0.025)
$p_2$	0.021	-0.033	-0.035
(Standard Deviation)	(0.025)	(0.026)	(0.021)
Number of instruments	11	11	11
Degrees of Freedom	6	6	6
J statistic	6.509	2.450	4.561
(Significance Level)	(0.368)	(0.874)	(0.601)
Implied <b>£</b>	-0.620	0.174	0.137
Implied $oldsymbol{J}$	2.854	-3.863	-2.530
H N	3555 18813	740 5029	413 2990

Table 2B Estimation and test results for the nonlinear Euler equation associated with the risky asset. Results obtained using instrument set two.

$$I = \boldsymbol{b} \left(\frac{c_{i,t+1}}{c_{i,t}}\right)^{y-1} \left(\frac{l_{i,t+1}}{l_{i,t}}\right)^{k} \left(\frac{p_{t+1} + d_{t+1}}{p_{t}}\right) \exp[\boldsymbol{p}_{1} fam_{i,t+1} + \boldsymbol{p}_{2} fam_{i,t}] + e_{i,t+1}$$

	All Households	Asset Holdings > 0	Asset Holdings > 1000
b	0.536	0.592	0.649
(Standard Deviation)	(0.139)	(0.156)	(0.240)
y - 1	-1.410	-2.538	-2.483
(Standard Deviation)	(0.762)	(0.626)	(1.009)
$\boldsymbol{k}$	5.181	-2.270	-3.032
(Standard Deviation)	(0.833)	(1.213)	(1.887)
$oldsymbol{p}_1$	0.015	0.089	0.069
(Standard Deviation)	(0.059)	(0.053)	(0.051)
$p_2$	0.009	-0.051	-0.059
(Standard Deviation)	(0.033)	(0.043)	(0.079)
Number of instruments	11	11	11
Degrees of Freedom	6	6	6
J statistic	15.610	8.605	19.470
(Significance Level)	(0.016)	(0.197)	(0.003)
Implied &	-0.086	0.403	0.328
Implied $oldsymbol{J}$	4.770	-3.809	-4.515
H N	3555 18813	740 5029	413 2990

Table 3A

Estimation and test results for the nonlinear Euler equation associated with the riskless asset. Results obtained using instrument set three.

$$I = \mathbf{b} \left(\frac{c_{i,t+1}}{c_{i,t}}\right)^{y-1} \left(\frac{l_{i,t+1}}{l_{i,t}}\right)^{\mathbf{k}} \left(\frac{1}{q_t}\right) \exp[\mathbf{p}_1 fam_{i,t+1} + \mathbf{p}_2 fam_{i,t}] + e_{i,t+1}$$

	All Households	Asset Holdings > 0	Asset Holdings > 1000
b	0.790	0.828	0.869
(Standard Deviation)	(0.098)	(0.084)	(0.057)
y - 1	-0.437	-1.731	-1.645
(Standard Deviation)	(0.421)	(0.420)	(0.374)
k	4.245	-1.267	-1.436
(Standard Deviation)	(1.045)	(0.670)	(0.735)
$oldsymbol{p}_1$	-0.011	0.050	0.038
(Standard Deviation)	(0.022)	(0.034)	(0.022)
$p_2$	0.013	-0.038	-0.035
(Standard Deviation)	(0.018)	(0.027)	(0.019)
Number of instruments	9	9	9
Degrees of Freedom	4	4	4
J statistic	3.091	13.235	6.768
(Significance Level)	(0.542)	(0.010)	(0.148)
Implied <b>£</b>	0.117	0.360	0.309
Implied $oldsymbol{J}$	4.808	-1.980	-2.081
H N	3555 18813	740 5029	413 2990

Table 3B Estimation and test results for the nonlinear Euler equation associated with the risky asset. Results obtained using instrument set three.

$$I = \boldsymbol{b} \left(\frac{c_{i,t+1}}{c_{i,t}}\right)^{y-1} \left(\frac{l_{i,t+1}}{l_{i,t}}\right)^{k} \left(\frac{p_{t+1} + d_{t+1}}{p_{t}}\right) \exp[\boldsymbol{p}_{1} fam_{i,t+1} + \boldsymbol{p}_{2} fam_{i,t}] + e_{i,t+1}$$

	All Households	Asset Holdings > 0	Asset Holdings > 1000
b	0.404	0.555	0.720
(Standard Deviation)	(0.068)	(0.191)	(0.225)
y - 1	-1.253	-2.749	-2.193
(Standard Deviation)	(0.707)	(0.767)	(1.103)
$\mathbf{k}$	6.445	-1.815	-2.085
(Standard Deviation)	(0.667)	(1.213)	(1.085)
$oldsymbol{p}_1$	0.013	0.010	0.054
(Standard Deviation)	(0.008)	(0.069)	(0.049)
$p_2$	0.019	-0.059	-0.045
(Standard Deviation)	(0.006)	(0.053)	(0.035)
Number of instruments	9	9	9
Degrees of Freedom	4	4	4
J statistic	2.456	9.664	13.228
(Significance Level)	(0.652)	(0.046)	(0.010)
Implied &	-0.040	0.490	0.363
Implied $oldsymbol{J}$	6.192	-3.564	-3.278
H N	3555 18813	740 5029	413 2990

Table 4A

Estimation and test results for the nonlinear Euler equation associated with the riskless asset. Results obtained using instrument set two. Time endowment set to 8760 hours.

Estimation and test results obtained by GMM estimation of the Euler equation

$$I = \boldsymbol{b} \left( \frac{c_{i,t+1}}{c_{i,t}} \right)^{y-1} \left( \frac{l_{i,t+1}}{l_{i,t}} \right)^{k} \left( \frac{1}{q_{t}} \right) \exp[\boldsymbol{p}_{1} fam_{i,t+1} + \boldsymbol{p}_{2} fam_{i,t}] + e_{i,t+1}$$

	All Households	Asset Holdings > 0	Asset Holdings > 1000
b	0.758	0.793	0.882
(Standard Deviation)	(0.090)	(0.079)	(0.075)
y - 1	-0.125	-1.644	-1.379
(Standard Deviation)	(0.656)	(0.472)	(0.391)
k	8.642	-5.661	-4.487
(Standard Deviation)	(1.780)	(1.265)	(1.870)
$p_1$	-0.022	0.040	0.034
(Standard Deviation)	(0.032)	(0.028)	(0.025)
$p_2$	0.028	-0.029	-0.034
(Standard Deviation)	(0.031)	(0.024)	(0.020)
Number of instruments	11	11	11
Degrees of Freedom	6	6	6
J statistic	17.455	3.096	4.019
(Significance Level)	(0.007)	(0.796)	(0.674)
Implied <b>£</b>	-0.014	0.102	0.077
Implied $oldsymbol{J}$	8.517	-6.306	-4.866
H N	3555 18813	740 5029	413 2990

Table 4B

Estimation and test results for the nonlinear Euler equation associated with the risky asset. Results obtained using instrument set two. Time endowment set to 8760 hours.

Estimation and test results obtained by GMM estimation of the Euler equation

$$I = \boldsymbol{b} \left(\frac{c_{i,t+1}}{c_{i,t}}\right)^{\mathbf{y}-1} \left(\frac{l_{i,t+1}}{l_{i,t}}\right)^{\mathbf{k}} \left(\frac{p_{t+1}+d_{t+1}}{p_{t}}\right) \exp[\boldsymbol{p}_{1} fam_{i,t+1} + \boldsymbol{p}_{2} fam_{i,t}] + e_{i,t+1}$$

	All Households	Asset Holdings > 0	Asset Holdings > 1000
b	0.556	0.589	0.638
(Standard Deviation)	(0.159)	(0.158)	(0.222)
y - 1	-1.490	-2.555	-2.514
(Standard Deviation)	(0.790)	(0.653)	(0.924)
k	9.065	-4.146	-5.930
(Standard Deviation)	(1.756)	(2.558)	(3.398)
$oldsymbol{p}_1$	0.019	0.090	0.069
(Standard Deviation)	(0.052)	(0.054)	(0.046)
$p_2$	0.010	-0.051	-0.057
(Standard Deviation)	(0.028)	(0.043)	(0.034)
Number of instruments	11	11	11
Degrees of Freedom	6	6	6
J statistic	14.151	8.115	21.848
(Significance Level)	(0.027)	(0.229)	(0.001)
Implied &	-0.057	0.272	0.203
Implied $oldsymbol{J}$	8.575	-5.702	-7.445
H N	3555 18813	740 5029	413 2990

Table 5A

Estimation and test results for the linearized Euler equation associated with the riskless asset. Results obtained using instrument set three.

Estimation and test results obtained by GMM estimation of

$$\Delta \ln(c_{i,t+1}) = \mathbf{w}_1 + \mathbf{w}_2 \ln(\frac{1}{q_t}) + \mathbf{w}_3 \Delta \ln(l_{i,t+1}) + \mathbf{p}_1 fam_{i,t+1} + \mathbf{p}_2 fam_{i,t} + u_{i,t+1}$$

	All Households	Asset Holdings > 0	Asset Holdings > 1000
<b>W</b> 1	0.035	0.032	0.038
(Standard Deviation)	(0.012)	(0.010)	(0.018)
<b>W</b> 2	0.148	0.224	0.178
(Standard Deviation)	(0.212)	(0.193)	(0.209)
<b>W</b> 3	-0.877	-1.659	-1.722
(Standard Deviation)	(0.549)	(0.426)	(0.885)
$oldsymbol{p}_1$	0.018	0.010	0.011
(Standard Deviation)	(0.005)	(0.006)	(0.012)
$p_2$	-0.027	-0.017	-0.020
(Standard Deviation)	(0.005)	(0.007)	(0.011)
Number of instruments	9	9	9
Degrees of Freedom	4	4	4
J statistic	43.187	53.327	16.724
(Significance Level)	(0.000)	(0.000)	(0.002)
Implied <b>y</b>	-5.756	-3.464	-4.617
Implied $m{k}$	-5.925	-7.406	-9.674
Implied <b>£</b>	0.492	0.318	0.323
Implied $oldsymbol{J}$	-11.682	-10.870	-14.292
H N	3555 18813	740 5029	413 2990

Table 5B

Estimation and test results for the linearized Euler equation associated with the risky asset. Results obtained using instrument set three.

Estimation and test results obtained by GMM estimation of

$$\Delta \ln(c_{i,t+1}) = \mathbf{w}_1 + \mathbf{w}_2 \ln(\frac{p_{t+1} + d_{t+1}}{p_t}) + \mathbf{w}_3 \Delta \ln(l_{i,t+1}) + \mathbf{p}_1 fam_{i,t+1} + \mathbf{p}_2 fam_{i,t} + u_{i,t+1}$$

	All Households	Asset Holdings > 0	Asset Holdings > 1000
<b>W</b> 1	0.032	0.033	0.039
(Standard Deviation)	(0.011)	(0.010)	(0.017)
<b>W</b> 2	0.071	0.059	0.036
(Standard Deviation)	(0.061)	(0.059)	(0.076)
<b>W</b> 3	-1.052	-1.797	-1.795
(Standard Deviation)	(0.446)	(0.470)	(0.847)
$p_1$	0.018	0.010	0.011
(Standard Deviation)	(0.005)	(0.006)	(0.012)
$p_2$	-0.028	-0.017	-0.020
(Standard Deviation)	(0.005)	(0.007)	(0.011)
Number of instruments	9	9	9
Degrees of Freedom	4	4	4
J statistic	481.434	41.537	16.375
(Significance Level)	(0.000)	(0.000)	(0.002)
Implied $oldsymbol{y}$	-13.084	-15.949	-26.777
Implied $m{k}$	-14.816	-30.457	-49.861
Implied &	0.468	0.343	0.349
Implied $oldsymbol{J}$	-27.901	-46.406	-76.638
H N	3555 18813	740 5029	413 2990

Table 6A

Estimation and test results for the nonlinear Euler equation associated with the riskless asset. Results obtained using instrument set four.

$$I = \boldsymbol{b} \left(\frac{c_{i,t+1}}{c_{i,t}}\right)^{\mathbf{y}-1} \left(\frac{l_{i,t+1}}{l_{i,t}}\right)^{\mathbf{k}} \left(\frac{1}{q_t}\right) \exp[\boldsymbol{p}_1 fam_{i,t+1} + \boldsymbol{p}_2 fam_{i,t}] + e_{i,t+1}$$

	All Households	Asset Holdings > 0	Asset Holdings > 1000
b	0.895	0.953	1.014
(Standard Deviation)	(0.141)	(0.095)	(0.119)
y - 1	-0.376	-1.163	-0.942
(Standard Deviation)	(0.371)	(0.233)	(0.339)
k	3.984	-0.202	-0.362
(Standard Deviation)	(1.085)	(1.201)	(1.533)
$p_1$	-0.010	0.021	0.009
(Standard Deviation)	(0.021)	(0.018)	(0.017)
$p_2$	0.013	-0.014	-0.008
(Standard Deviation)	(0.017)	(0.015)	(0.018)
$d_1$	-0.005	0.0003	-0.001
(Standard Deviation)	(0.010)	(0.0040)	(0.005)
$d_2$	0.0006	-0.00003	-0.067
(Standard Deviation)	(0.0013)	(0.00006)	(0.785)
Number of instruments	11	11	11
Degrees of Freedom	4	4	4
J statistic	5.521	10.049	27.387
(Significance Level)	(0.237)	(0.039)	(0.000)
Implied <b>£</b>	0.135	0.039	-0.190
Implied $oldsymbol{J}$	4.608	-4.179	-0.304
H N	3555 18813	740 5029	413 2990

Table 6B Estimation and test results for the nonlinear Euler equation associated with the risky asset. Results obtained using instrument set four.

$$I = \boldsymbol{b} \left(\frac{c_{i,t+1}}{c_{i,t}}\right)^{\mathbf{y}-1} \left(\frac{l_{i,t+1}}{l_{i,t}}\right)^{\mathbf{k}} \left(\frac{p_{t+1}+d_{t+1}}{p_{t}}\right) \exp[\boldsymbol{p}_{1} fam_{i,t+1} + \boldsymbol{p}_{2} fam_{i,t}] + e_{i,t+1}$$

	All Households	Asset Holdings > 0	Asset Holdings > 1000
b	1.472	0.442	0.758
(Standard Deviation)	(1.006)	(0.361)	(0.476)
<b>y</b> −1	-1.191	-2.540	-1.925
(Standard Deviation)	(0.734)	(0.946)	(1.315)
k	6.496	-1.030	-1.471
(Standard Deviation)	(0.668)	(2.558)	(1.418)
$p_1$	0.003	0.084	0.034
(Standard Deviation)	(0.077)	(0.075)	(0.047)
$p_2$	0.041	-0.051	-0.028
(Standard Deviation)	(0.054)	(0.055)	(0.039)
$d_1$	-0.071	0.021	0.003
(Standard Deviation)	(0.046)	(0.029)	(0.021)
$d_2$	0.090	-0.00029	-0.00007
(Standard Deviation)	(0.056)	(0.00038)	(0.00020)
Number of instruments	11	11	11
Degrees of Freedom	4	4	4
J statistic	6.133	11.754	16.639
(Significance Level)	(0.189)	(0.019)	(0.002)
Implied £	-0.030	0.599	0.386
Implied $oldsymbol{J}$	6.305	-2.570	-2.396
H N	3555 18813	740 5029	413 2990

Table 7A

Estimation and test results for the nonlinear Euler equation associated with the riskless asset. Results obtained using instrument set five.

$$I = \mathbf{b} \left( \frac{c_{i,t+1}}{c_{i,t}} \right)^{\mathbf{y}-1} \left( \frac{l_{i,t+1}}{l_{i,t}} \right)^{\mathbf{k}} \left( \frac{1}{q_t} \right) \exp[\mathbf{p}_1 fam_{i,t+1} + \mathbf{p}_2 fam_{i,t}] + e_{i,t+1}$$

	All Households	Asset Holdings > 0	Asset Holdings > 1000
b	0.892	0.430	0.615
(Standard Deviation)	(0.087)	(0.337)	(0.303)
<b>y</b> −1	-0.994	-2.342	-1.972
(Standard Deviation)	(0.560)	(0.813)	(0.548)
k	0.146	-2.211	-2.225
(Standard Deviation)	(1.064)	(0.796)	(0.714)
$oldsymbol{p}_1$	0.027	0.079	0.053
(Standard Deviation)	(0.024)	(0.062)	(0.030)
$p_2$	-0.022	-0.047	-0.046
(Standard Deviation)	(0.016)	(0.042)	(0.025)
$d_1$	0.0018	0.050	0.024
(Standard Deviation)	(0.0040)	(0.078)	(0.091)
$d_2$	0.0007	-0.0012	-0.0004
(Standard Deviation)	(0.0019)	(0.0026)	(0.0033)
Number of instruments	11	11	11
Degrees of Freedom	4	4	4
J statistic	29.695	0.881	5.809
(Significance Level)	(0.000)	(0.927)	(0.213)
Implied <b>£</b>	0.039	0.377	0.304
Implied $oldsymbol{J}$	0.152	-3.553	-3.197
H N	3555 18813	740 5029	413 2990

Estimation and test results for the nonlinear Euler equation associated with the risky asset. Results obtained using instrument set five.

Table 7B

Estimation and test results obtained by GMM estimation of the Euler equation

$$I = \boldsymbol{b} \left(\frac{c_{i,t+1}}{c_{i,t}}\right)^{\mathbf{y}-1} \left(\frac{l_{i,t+1}}{l_{i,t}}\right)^{\mathbf{k}} \left(\frac{p_{t+1}+d_{t+1}}{p_{t}}\right) \exp[\boldsymbol{p}_{1} fam_{i,t+1} + \boldsymbol{p}_{2} fam_{i,t}] + e_{i,t+1}$$

	All Households	Asset Holdings > 0	Asset Holdings > 1000
b	0.458	1.006	0.567
(Standard Deviation)	(0.178)	(0.257)	(0.509)
<b>y</b> −1	-1.228	-3.556	-2.826
(Standard Deviation)	(0.631)	(0.900)	(1.394)
k	6.649	-2.777	-2.630
(Standard Deviation)	(0.737)	(1.416)	(1.359)
$oldsymbol{p}_1$	0.012	0.155	0.083
(Standard Deviation)	(0.077)	(0.101)	(0.076)
$p_2$	0.020	-0.064	-0.061
(Standard Deviation)	(0.062)	(0.089)	(0.049)
$d_1$	-0.020	0.153	-0.016
(Standard Deviation)	(0.049)	(0.299)	(0.192)
$d_2$	0.005	-0.0043	0.0012
(Standard Deviation)	(0.023)	(0.0102)	(0.0072)
Number of instruments	11	11	11
Degrees of Freedom	4	4	4
J statistic	1.964	2.543	11.063
(Significance Level)	(0.742)	(0.636)	(0.025)
Implied <b>£</b>	-0.035	0.479	0.409
Implied $oldsymbol{J}$	6.421	-5.333	-4.456
H N	3555 18813	740 5029	413 2990

Table 8A

Estimation and test results for the linearized Euler equation associated with the riskless asset. Results obtained using instrument set four.

Estimation and test results obtained by GMM estimation of

$$\Delta \ln(c_{i,t+1}) = \mathbf{w}_1 + \mathbf{w}_2 \ln(\frac{1}{q_t}) + \mathbf{w}_3 \Delta \ln(l_{i,t+1}) + \mathbf{p}_1 fam_{i,t+1} + \mathbf{p}_2 fam_{i,t} + u_{i,t+1}$$

	All Households	Asset Holdings > 0	Asset Holdings > 1000
<b>W</b> 1	0.091	0.079	0.101
(Standard Deviation)	(0.038)	(0.073)	(0.115)
<b>W</b> 2	0.301	0.301	0.243
(Standard Deviation)	(0.295)	(0.211)	(0.243)
<b>W</b> 3	0.167	-0.155	-1.020
(Standard Deviation)	(0.893)	(0.867)	(2.251)
$p_1$	0.012	0.004	0.003
(Standard Deviation)	(0.004)	(0.008)	(0.020)
$p_2$	-0.019	-0.008	-0.009
(Standard Deviation)	(0.004)	(0.008)	(0.021)
$d_1$	-0.0014	-0.001	-0.0019
(Standard Deviation)	(0.0020)	(0.003)	(0.0054)
$d_2$	-0.000008	-0.000012	0.000002
(Standard Deviation)	(0.000026)	(0.000043)	(0.000068)
Number of instruments	11	11	11
Degrees of Freedom	4	4	4
J statistic	27.339	2.989	3.951
(Significance Level)	(0.000)	(0.559)	(0.412)
Implied $oldsymbol{y}$	-2.322	-2.322	-3.115
Implied ${m k}$	0.544	-0.514	-4.197
Implied <b>£</b>	1.313	0.818	0.425
Implied $oldsymbol{J}$	-1.767	-2.837	-7.312
H N	3555 18813	740 5029	413 2990

Table 8B

Estimation and test results for the linearized Euler equation associated with the risky asset. Results obtained using instrument set four.

Estimation and test results obtained by GMM estimation of

$$\Delta \ln(c_{i,t+1}) = \mathbf{w}_1 + \mathbf{w}_2 \ln(\frac{p_{t+1} + d_{t+1}}{p_t}) + \mathbf{w}_3 \Delta \ln(l_{i,t+1}) + \mathbf{p}_1 fam_{i,t+1} + \mathbf{p}_2 fam_{i,t} + u_{i,t+1}$$

	All Households	Asset Holdings > 0	Asset Holdings > 1000
<b>W</b> 1	0.070	0.063	0.094
(Standard Deviation)	(0.040)	(0.075)	(0.118)
<b>W</b> 2	0.088	0.068	0.050
(Standard Deviation)	(0.068)	(0.061)	(0.072)
<b>W</b> 3	-0.435	-0.611	-1.503
(Standard Deviation)	(0.525)	(0.780)	(1.974)
$p_1$	0.013	0.004	0.006
(Standard Deviation)	(0.004)	(0.007)	(0.020)
$p_2$	-0.021	-0.009	-0.013
(Standard Deviation)	(0.004)	(0.008)	(0.020)
Number of instruments	11	11	11
Degrees of Freedom	6	6	4
J statistic	20.212	16.620	30.239
(Significance Level)	(0.000)	(0.000)	(0.000)
Implied $oldsymbol{y}$	-10.363	-13.705	-19.000
Implied <b>K</b>	-4.943	-8.905	-30.060
Implied &	0.677	0.604	0.387
Implied $oldsymbol{J}$	-15.306	-22.691	-49.060
H N	3555 18813	740 5029	413 2990

Table 9A

Estimation and test results for the linearized Euler equation associated with the riskless asset. Results obtained using instrument set five.

Estimation and test results obtained by GMM estimation of

$$\Delta \ln(c_{i,t+1}) = \mathbf{w}_1 + \mathbf{w}_2 \ln(\frac{1}{q_t}) + \mathbf{w}_3 \Delta \ln(l_{i,t+1}) + \mathbf{p}_1 fam_{i,t+1} + \mathbf{p}_2 fam_{i,t} + u_{i,t+1}$$

	All Households	Asset Holdings > 0	Asset Holdings > 1000
<b>W</b> 1	0.001	-0.186	-0.055
(Standard Deviation)	(0.029)	(0.138)	(0.325)
<b>W</b> 2	0.045	0.191	0.166
(Standard Deviation)	(0.187)	(0.198)	(0.230)
<b>W</b> 3	-1.672	-2.413	-2.484
(Standard Deviation)	(0.556)	(0.489)	(0.581)
$p_1$	0.019	0.009	0.014
(Standard Deviation)	(0.006)	(0.007)	(0.014)
$p_2$	-0.028	-0.016	-0.023
(Standard Deviation)	(0.006)	(0.008)	(0.013)
$d_1$	0.002	0.029	0.011
(Standard Deviation)	(0.004)	(0.020)	(0.046)
$d_2$	0.00002	-0.0009	-0.0003
(Standard Deviation)	(0.00016)	(0.0007)	(0.0016)
Number of instruments	11	11	11
Degrees of Freedom	4	4	4
J statistic	29.002	71.707	231.738
(Significance Level)	(0.000)	(0.000)	(0.000)
Implied $oldsymbol{y}$	-21.222	-4.235	-5.024
Implied ${m k}$	-37.155	-12.633	-14.963
Implied <b>£</b>	0.363	0.251	0.251
Implied $oldsymbol{J}$	-58.377	-16.869	-19.987
H N	3555 18813	740 5029	413 2990

Table 9B

Estimation and test results for the linearized Euler equation associated with the risky asset. Results obtained using instrument set five.

Estimation and test results obtained by GMM estimation of

$$\Delta \ln(c_{i,t+1}) = \mathbf{w}_1 + \mathbf{w}_2 \ln(\frac{p_{t+1} + d_{t+1}}{p_t}) + \mathbf{w}_3 \Delta \ln(l_{i,t+1}) + \mathbf{p}_1 fam_{i,t+1} + \mathbf{p}_2 fam_{i,t} + u_{i,t+1}$$

	All Households	Asset Holdings > 0	Asset Holdings > 1000
<b>W</b> 1	-0.001	-0.192	-0.055
(Standard Deviation)	(0.030)	(0.145)	(0.329)
<b>W</b> 2	0.062	0.059	0.039
(Standard Deviation)	(0.057)	(0.063)	(0.089)
<b>W</b> 3	-1.658	-2.510	-2.530
(Standard Deviation)	(0.501)	(0.587)	(0.582)
$oldsymbol{p}_1$	0.019	0.009	0.014
(Standard Deviation)	(0.006)	(0.007)	(0.014)
$p_2$	-0.027	-0.016	-0.023
(Standard Deviation)	(0.006)	(0.008)	(0.013)
Number of instruments	11	11	11
Degrees of Freedom	4	4	4
J statistic	32.623	12.393	25.492
(Significance Level)	(0.000)	(0.014)	(0.000)
Implied $oldsymbol{y}$	-15.129	-15.949	-24.641
Implied <b>k</b>	-26.741	-42.524	-64.871
Implied &	0.361	0.272	0.275
Implied $oldsymbol{J}$	-41.870	-58.491	-89.512
H N	3555 18813	740 5029	413 2990